Cross-Border Equity Holdings and External Accounts^{*}

Anna Pavlova London Business School and CEPR Roberto Rigobon Sloan School of Management, MIT and NBER

This draft: July 2015

Abstract

Recent evidence on the importance of cross-border equity flows calls for a rethinking of the standard theory of external adjustment. We introduce equity holdings and portfolio choice into an otherwise conventional open-economy dynamic equilibrium model. Our model is simple and it admits an exact closed-form solution regardless of whether financial markets are complete or incomplete. In this simple framework, we are able to establish interconnections between the real side of the economy, represented by the trade balance, the current account, and consumption allocations and the financial side, such as portfolio holdings, stock prices and valuation changes. Methodologically, we contribute to the existing DSGE literature on international portfolio choice that so far has either relied on approximation methods or used simplified asset market structures.

JEL Classifications: G12, G15, F31, F36

Keywords: Current account, portfolio rebalancing, capital gains, international finance, asset pricing, global imbalances.

^{*}E-mail addresses: apavlova@london.edu and rigobon@mit.edu. This paper was previously circulated under the title "Equilibrium Portfolios and External Adjustment under Incomplete Markets." We thank Philippe Bacchetta, Mick Devereux, Bernard Dumas, Pierre-Olivier Gourinchas, Robert Kollmann, Enrique Mendoza, Stavros Panageas, Hélène Rey, Carlos Végh and seminar participants at the Bank of England, Berkeley, British Columbia, Cornell, Dartmouth, Imperial, Kellogg, Lausanne, LBS, Maryland, Michigan, Prague, PUC-Rio, San Francisco Fed, Yale and various conferences for stimulating discussions. Pavlova thanks the European Research Council for financial support (Grant ERC263312).

1. Introduction

Since David Hume's seminal work in 1752, "Of the Balance of Trade", economists have been confronted with the question of external account sustainability. The issue, in principle, is a simple one: are a country's sources of income today and in the future capable of financing its consumption plan? In practice, however, the answer has been far more elusive. In recent years, the swelling of cross-border equity holdings and capital gains on these holdings has complicated the matter even further. The recent empirical literature has argued that these capital gains, currently misreported in the national accounts because of the lack of marking to market, could account for most of the required adjustment in developed nations during the last decade. Although more work lies ahead, the empirical literature has advanced significantly. The theoretical counterpart, on the other hand, has encountered several obstacles. First, most open economy macroeconomics models abstract from equity markets and portfolio choice, which makes them inadequate for studying capital gains. Second, the few models that do take into consideration cross-country holdings typically require that financial markets are complete. In his Ohlin lecture, Maurice Obstfeld remarks that "Portfolio choice under incomplete markets is largely terra incognita". A handful of recently-developed models that make progress in this direction rely on new approximation methods that are still in the process of being developed.

In this paper, we offer a simple framework in which we incorporate portfolio choice and asset pricing in an otherwise standard open economy dynamic equilibrium model. We are able to describe asset prices, exchange rates, portfolio holdings, capital gains, and external accounts exactly, in closed form, regardless of whether financial markets are complete or incomplete. Our model allows us to understand the interconnections between portfolio decisions and the measures of external imbalances in the international economics literature. For example, in our model, a country that holds a riskier external portfolio vis-à-vis the rest of the world would enjoy a risk premium, which is used towards financing its trade deficit. A country like this can potentially run a persistent trade deficit and a current account deficit, effectively receiving wealth transfers from the rest of the world. Of course, during financial crises, such a country suffers more severe losses on its portfolio relative to the rest of the world. As Gourinchas and Rey (2007b) and Gourinchas, Govillot, and Rey (2010) document, this description fits the US, whose external portfolio is indeed riskier than that of the rest of the world, with its assets dominated by risky equity-type assets and the liabilities by safe government bonds. The US has been running a current account deficit for the last three decades, which has long puzzled many commentators. Our framework allows us to rationalize this fact and to also shed some light on conditions under which a particular risk profile is optimal for a country, how a country's external accounts respond to a shock in financial markets, and whether market incompleteness matters.

In our model, each country is endowed with a Lucas tree, producing a country-specific good. Each tree's output is driven by its own supply shock, and stocks are claims to the Lucas trees. There is also a bond that provides opportunities for riskless borrowing and lending. Each country is populated by a representative consumer/investor who has log-linear preferences over the two goods. Uncertainty in the economy is due to output shocks in each country and to the consumers' demand shocks. Each demand shock reflects a shift in preferences towards a particular good, as in Dornbusch, Fischer, and Samuelson (1977). These demand shocks are a convenient source of heterogeneity in cross-country appetites for risky assets as they give rise to hedging portfolios that are different across countries. There are more shocks than there are assets, and so markets are incomplete. In this model, we define the capital-gains adjusted current account, given by a change in the net foreign asset position of a country. Since the countries' hold long-lived assets (stocks), this notion of the current account differs from the conventional current account (as reported in the national accounts) in that it incorporates capital gains on Net Foreign Assets (NFA) positions (the so-called valuation effects).

Our model allows us to revisit several important relationships and mechanisms that underpin the traditional intertemporal approach to the current account. First, in contrast to the traditional view, it is entirely possible in our model for a debtor country to run long persistent current account deficits, including running a current account deficit forever. This is possible because we allow for cross-border equity holdings, and (positive) equity holdings of a country that is a net borrower in the international debt market can finance the current account deficits. Second, we find that there is a one-to-one mapping between the trade balance, a traditional measure of external imbalances based on real variables, and the size of the NFA position, a measure of external imbalances based on financial holdings. Any shock to an NFA position of a country must then be accompanied by an offsetting adjustment of the trade balance; otherwise the consumption profile becomes unaffordable. This is an equilibrium relationship that holds under both complete and incomplete markets. A similar result has been obtained in traditional bonds-only models of international economics; it turns out to be robust to the introduction of cross-border equity trade. Third, we establish that the response of NFA to real shocks depends on the (endogenous) correlation of the cross-country stock returns and on the optimal portfolios of the countries. Our analysis demonstrates that the direction of the adjustment of the current account following a shock depends on whether a country is a net creditor or net debtor. For example, applied to the recent financial crisis, our model predicts that the drop in output in the US and UK reduces these countries' current account deficits, while the exact same drop in output in China and Japan reduces their surpluses. While the current accounts improve for the first group of countries and deteriorate for the second, however, the stock prices of all countries fall in tandem. Our model highlights that when the correlations of stock market returns are endogenous and portfolio holdings are determined in equilibrium, the question of how external accounts change after a particular shock cannot be answered without understanding the dynamics of asset returns and the compositions of the international portfolios.

1.1. Related Literature

Our analysis is motivated by the recent empirical literature that has documented the importance of valuation changes (Gourinchas and Rey (2007a), Gourinchas and Rey (2007b), Lane and Milesi-Ferretti (2001), Lane and Milesi-Ferretti (2007), Tille (2003), and Tille (2008)). This literature emphasizes that capital gains on countries' NFA positions are large, volatile, serially uncorrelated, and effectively dominate the movements in NFA (Kollmann (2006) and Lane and Shambaugh (2010)). Moreover, these capital gains are negatively correlated with the trade balance, and so capital gains stabilize countries' NFA positions by offsetting trade balance movements (Gourinchas and Rey (2007b), Devereux and Sutherland (2010)). Our model has the time series implications that are consistent with this empirical evidence. However, exploring these implications is beyond the scope of this paper. We undertake this exercise in a companion paper, Pavlova and Rigobon (2010), whose main goal is to calibrate a simpler version of the model developed here and to apply it to study several questions within the global imbalances literature. In contrast, the goal of this paper is to make progress on the "Obstfeld's agenda" (incomplete markets) by establishing a methodology for solving international portfolio models under incomplete markets. This methodology does not cover just the case of incomplete (missing) markets. Another natural application of our methodology is to the case of restricted participation or other types of portfolio constraints.

Our work belongs to the strand of literature that introduces valuation changes in models of in-

ternational macroeconomics. This literature relies primarily on approximation methods. Ghironi, Lee, and Rebucci (2006) and Kollmann (2006) compute portfolios and NFA changes using standard first-order approximations around a deterministic steady state. The second-generation methodologies that grow out of Samuelson (1970) and Judd and Guu (2001) were developed by Devereux and Sutherland (2011), Engel and Matsumoto (2009), Evans and Hnatkovska (2007), and Tille and van Wincoop (2010), who employ higher-order approximations to analyze countries' portfolios and external accounts. To this day, little is known about the behavior of these economies away from the deterministic steady state, where the underlying volatilities are not small.¹ By contrast, our model admits an exact analytical solution, even under incomplete markets, and as such can serve as a natural benchmark to assess the accuracy of these popular approximation methods.

In terms of the modeling framework, our paper is closely related to Pavlova and Rigobon (2007) and Pavlova and Rigobon (2008), both of which are analytically tractable international macro-finance models. The main difference from these papers is that neither of them examines external adjustment. In fact, in those models the current account equals to zero. We extend their framework to produce meaningful implications for the current account. On the technical side, this is an extension of their framework to incomplete markets, which is non-trivial and thus constitutes a methodological contribution of this paper. The fact that in the real world markets are incomplete hardly requires any motivation. If markets were complete and all risks could be perfectly hedged, there would be no role for a policy response and the subject of external imbalances would be purely academic. Our model is also related to Cole and Obstfeld (1991), Helpman and Razin (1978), and Zapatero (1995). All these are tractable multi-asset multi-good models like ours, but in contrast to our work, in each of these papers markets are complete or effectively complete. Our paper complements Kraay and Ventura (2000) and Kraay and Ventura (2003), who assume a flat supply of capital, eliminating valuation effects and focusing on investment. We employ an endowment economy, and hence investment is not modeled, and focus entirely on valuation effects.² Finally, two recent papers that embark on an agenda very close to ours are Gourinchas, Govillot, and Rey (2010) and Maggiori (2012). Like us, they consider two-country endowment economies and, like

¹More generally, the accuracy and performance and of numerical algorithms for solving multi-country dynamic stochastic general equilibrium (DSGE) models remain the subject of an ongoing debate. The complexity of the problem makes some of the algorithms quite inaccurate, especially away from the steady state. An excellent reference for the current state of this debate is a recent special issue of the Journal of Economic Dynamics and Control on numerical methods for multi-country DSGE models (see Den Haan, Judd, and Juillard (2010)).

 $^{^{2}}$ Also related, but, unlike ours, cast in the context of a production economy, is the elegant analysis of Devereux and Saito (2006).

us, link persistent current account deficits to asymmetric risk sharing between the two countries. Asymmetries between countries in these papers stem from differences in financial development. The papers have appeared after the first draft of our paper was circulated, and they lend strong support to our framework and our intuitions.

In terms of the methodology, our paper is closely related to He and Pearson (1991), Cuoco and He (1994), and Basak and Cuoco (1998). At a partial equilibrium level, He and Pearson derive a solution to a consumption-portfolio problem under incomplete markets.³ Our contribution is to embed their technique in a general equilibrium setting and establish the effects of incomplete markets on asset prices. Cuoco and He develop a method for solving for equilibrium under incomplete markets via a "planner" with stochastic weights. Basak and Cuoco were the first to apply this method to study financial markets with frictions (restricted participation, in their case). None of these two papers, however, offers a model with multiple risky assets and incomplete markets that can be analyzed analytically.

The rest of the paper is organized as follows. Section 2 describes the model and characterizes its equilibrium. Section 3 derives a number of implications of our model for the current account and its dynamics. Section 4 presents several special cases of our economy in which the characterization of the countries' portfolios simplify significantly. Section 5 highlights the external adjustment mechanism in our model, Section 6 discusses several caveats and desirable extensions, and Section 7 concludes.

2. The Model

2.1. The Economic Setting

We consider a continuous-time pure-exchange world economy with a finite horizon, [0, T] along the lines of Pavlova and Rigobon (2007).⁴ The main advantage of using continuous time is tractability an analogous discrete-time model does not admit a closed-form solution. Uncertainty is represented by a filtered probability space $(\Omega, \mathcal{F}, \{\mathcal{F}_t\}, P)$, on which is defined a standard four-dimensional Brownian motion $\vec{w}(t) = (w(t), w^*(t), w^{\gamma}, w^{\theta})^{\top}, t \in [0, T]$. All stochastic processes are assumed adapted to $\{\mathcal{F}_t; t \in [0, T]\}$, the augmented filtration generated by \vec{w} . All stated (in)equalities

³See Schroder and Skiadas (2003) for a related analysis under a more general class of preferences.

⁴The assumption of finite horizon is for simplicity. We rely on existence results and regularity conditions developed in asset pricing for finite horizon models. T can be arbitrarily large but finite.

involving random variables hold *P*-almost surely. In what follows, given our focus, we assume all processes introduced to be well-defined, without explicitly stating regularity conditions ensuring this.

There are two countries in the world economy: Home and Foreign. The Home country represents a large industrialized country, while Foreign stands for the rest of the world. Each country is endowed with a Lucas tree producing a strictly positive amount of a country-specific perishable good, Y and Y^* , respectively:

$$dY(t) = \mu_Y(t) Y(t) dt + \sigma_Y(t) Y(t) dw(t)$$
 (Home), (1)

$$dY^{*}(t) = \mu_{Y^{*}}(t) Y^{*}(t) dt + \sigma_{Y^{*}}(t) Y^{*}(t) dw^{*}(t)$$
 (Foreign), (2)

where μ_Y , μ_Y^* , $\sigma_Y > 0$, and $\sigma_Y^* > 0$ are arbitrary adapted processes. The claims to the trees, Home and Foreign stocks S and S^* , respectively, are available for trade by all investors and are in fixed supply of one share each. The prices of the Home and Foreign goods are denoted by p and p^* , respectively. We fix the world numeraire basket to contain $a \in (0, 1)$ units of the Home good and (1-a) units of the Foreign good, and normalize the price of this basket to be equal to unity. The terms of trade, q, are defined as the price of the Home good relative to that of the Foreign good: $q \equiv p/p^*$.

In addition to the stocks S and S^* , there is also the "world" bond with the price B available for investment, which is a money market account locally riskless in units of the numeraire. The bond is in zero net supply. Since there are four independent Brownian motions driving the economy and only three investment opportunities in place, financial markets are incomplete. To fix notation, the posited dynamics of the investment opportunity set of the agents is given by

$$dB(t) = B(t)r(t)dt, (3)$$

$$dS(t) + p(t)Y(t)dt = S(t)[\mu_S(t)dt + \sigma_S(t)d\vec{w}(t)], \qquad (4)$$

$$dS^{*}(t) + p^{*}(t)Y^{*}(t)dt = S^{*}(t)[\mu_{S^{*}}(t)dt + \sigma_{S^{*}}(t)d\vec{w}(t)],$$
(5)

where the interest rate r, the stocks expected returns μ_s and μ_{s^*} and their volatilities σ_s and σ_{s^*} are to be determined in equilibrium. The *volatility matrix* of the stock returns is then defined as $\sigma \equiv \begin{bmatrix} \sigma_s \\ \sigma_{s^*} \end{bmatrix}.$

The initial shareholdings of a representative consumer-investor of each country consist of no shares of the bond and a total supply of the stock market of his country. Thus, the initial wealth of the Home resident is $W_H(0) = S(0)$ and that of the Foreign resident is $W_F(0) = S^*(0)$. Each consumer $i, i \in \{H, F\}$, chooses nonnegative consumption of each good $(C_i(t), C_i^*(t))$ and a portfolio of the available securities $x_i(t) \equiv (x_i^S(t), x_i^{S^*}(t))$, where x_i^j denotes the fraction of wealth of consumer *i* invested in asset *j*. The dynamic budget constraint of each consumer has the standard form

$$dW_{i}(t) = \left[W_{i}(t)r(t) + x_{i}^{S}(t)W_{i}(t)(\mu_{S}(t) - r(t)) + x_{i}^{S^{*}}(t)W_{i}(t)(\mu_{S^{*}}(t) - r(t))\right]dt + \left[x_{i}^{S}(t)W_{i}(t)\sigma_{S}(t) + x_{i}^{S^{*}}(t)W_{i}(t)\sigma_{S^{*}}(t)\right]d\vec{w}(t) - \left[p(t)C_{i}(t) + p^{*}(t)C_{i}^{*}(t)\right]dt, \quad (6)$$

where $W_i(T) \ge 0$, $i \in \{H, F\}$. Preferences of consumer *i*, are represented by a time-additive log-linear utility function defined over consumption of both goods:⁵

$$E\left[\int_{0}^{T} e^{-\rho t} u_{i}(C_{i}(t), C_{i}^{*}(t)) dt\right], \qquad \rho > 0, \quad i \in \{H, F\},$$
(7)

where

$$u_{H}(C_{H}(t), C_{H}^{*}(t)) = \gamma_{H}(t) \left[\theta_{H}(t) \log C_{H}(t) + (1 - \theta_{H}(t)) \log C_{H}^{*}(t) \right]$$
$$u_{F}(C_{F}(t), C_{F}^{*}(t)) = \gamma_{F} \left[(1 - \theta_{F}) \log C_{F}(t) + \theta_{F} \log C_{F}^{*}(t) \right].$$

In our specification of the countries' utilities, we allow for the possibility of preference shifts towards the home or the foreign good (or "demand shocks"), modeled along the lines of Dornbusch, Fischer, and Samuelson (1977). There are several reasons why we consider demand shocks. First, in the absence of the demand shocks, free trade in goods makes stock prices perfectly correlated and financial markets irrelevant (Helpman and Razin (1978), Cole and Obstfeld (1991), Zapatero (1995)). Second, empirical evidence indicates that demand shocks are important drivers of the real-world dynamics. For example, Stockman and Tesar (1995) argue that preference shocks are roughly 85% of the size of supply shocks. Third, these demand shocks represent a convenient modeling device that allows to analyze international differences in appetite for risky assets in a very tractable way. For example, our model is more tractable than the one in which the two countries differ in their risk tolerance, but it can deliver similar implications. As argued in Pavlova and

⁵Our choice of logarithmic preferences is dictated by tractability; otherwise, we do not obtain closed-form solutions for stock prices and portfolios. In Section 6 we discuss how one may try to relax this assumption. It is well-known that these preferences give rise to small equity risk premia. While it is not our goal in this paper to match the assetpricing moments, an extension of our model that does so is clearly desirable. Some promising recent work along this dimension has been done by Stathopoulos (2008) who extends our framework and analyzes a two-country two-good model in which agents have logarithmic preferences with external habits (see Section 6 for more discussion).

Rigobon (2007), the demand shocks can have a broad range of interpretations such as differences of opinion, catching up with the Joneses, or pure demand shifts.

The stochastic term γ_H , as we demonstrate below, gives rise to intertemporal hedging motives. The stochastic term θ_H represents an intratemporal demand shift towards a particular good. As we are going to show below, it does not give rise to any additional hedging portfolios, which is why we find it useful to separate the component of the demand shock that operates across time (γ_H) from the component that operates across goods (θ_H). Formally, we assume that γ_H and θ_H follow

$$d\gamma_H(t) = \sigma_{\gamma_H}(t)d\vec{w}(t), \quad d\theta_H(t) = \mu_{\theta_H}(t)dt + \sigma_{\theta_H}(t)d\vec{w}(t).$$

We additionally require that γ_H and $\gamma_H \times \theta_H$ are strictly positive adapted stochastic processes and martingales. In the analysis that follows, we consider primarily two types of demand shocks: (i) demand shocks that are completely independent of the supply (output) shocks w and w^* and (ii) demand shocks that are allowed to be correlated with the supply shocks. We assume that there are no demand shocks at Foreign. This asymmetry does not limit the ability of our model to produce a wide range of risk-sharing patterns: there is enough heterogeneity in this simpler version, and the extended version would not yield many additional insights. Additionally, this specification is better for analytical tractability. If we were to introduce demand shocks at Foreign, we can solve the model without relying on approximations only in some special cases.

2.2. Countries' Portfolio Choice

The first step in our solution procedure is to derive the countries' optimal portfolios at a partial equilibrium level. To do so, we are going to employ techniques developed in the portfolio choice literature. However, relative to that literature, there are two non-standard ingredients in the optimization problem that the countries are facing: multiple consumption goods and incomplete markets. We address them in turn.

For concreteness, we focus our exposition on the Home consumer. The portfolio of the Foreign consumer is derived analogously. Following the early literature in finance (Breeden (1979), Adler and Dumas (1983)), we decompose the problem of maximizing his utility (7) subject to the budget constraint (6) into two parts. First, at each t, we derive the consumer's demands for the Home and the Foreign goods, keeping the overall consumption expenditure fixed. Second, we derive his optimal consumption expenditure process and the optimal portfolio.

The first step is the standard static consumer problem under certainty:

$$\max_{\{C_H(t), C_H^*(t)\}} \gamma_H(t) \left[\theta_H(t) \log C_H(t) + (1 - \theta_H(t)) \log C_H^*(t) \right]$$

s.t. $p(t)C_H(t) + p^*(t)C_H^*(t) \le \mathbb{C}_H(t),$

where $\mathbb{C}_{H}(t)$ denotes overall consumption expenditure at time t. Solving this problem, we obtain the following demands for the individual goods as fractions of the overall expenditure:

$$\overline{C}_{H}(t) = \theta_{H}(t) \frac{\mathbb{C}_{H}(t)}{p(t)}, \quad \overline{C}_{H}^{*}(t) = (1 - \theta_{H}(t)) \frac{\mathbb{C}_{H}(t)}{p^{*}(t)}.$$
(8)

The indirect utility function defined as $U_H(\mathbb{C}_H(t); p(t), p^*(t)) \equiv u_H(\overline{C}_H(t), \overline{C}_H^*(t))$ is then given by

$$U_{H}(\mathbb{C}_{H}(t); p(t), p^{*}(t)) = \gamma_{H}(t) \log \mathbb{C}_{H}(t) + F(\gamma_{H}(t), p(t), p^{*}(t)),$$

where $F(\cdot)$ is a function the form of which does not affect our analysis. This function F depends only on the variables that are exogenous from the viewpoint of the consumer and therefore, because of the separability of the indirect utility, it drops out of his portfolio choice.

The second step is to reformulate the portfolio choice problem of the consumer in terms of his indirect utility:

$$\max_{x_H^S, x_H^{S^*}, \mathbb{C}_H} E\left[\int_0^T e^{-\rho t} \gamma_H(t) \log \mathbb{C}_H(t) dt\right]$$
(9)

s.t.
$$dW_{H}(t) = \left[W_{H}(t) r(t) + x_{H}^{S}(t) W_{H}(t) (\mu_{S}(t) - r(t)) + x_{H}^{S^{*}}(t) W_{H}(t) (\mu_{S^{*}}(t) - r(t)) \right] dt \\ + \left[x_{H}^{S}(t) W_{H}(t) \sigma_{S}(t) + x_{H}^{S^{*}}(t) W_{H}(t) \sigma_{S^{*}}(t) \right] d\vec{w}(t) - \mathbb{C}_{H}(t) dt .$$
(10)

The optimization problem is thus formally equivalent to a familiar single-good consumption-investment problem, with consumption expenditure \mathbb{C}_H replacing consumption. Consumption of individual goods can then be recovered from (8). It is important to note that the prices of the individual goods, p and p^* , and hence the terms of trade have dropped out of the optimization problem. This implies that fluctuations in the terms of trade do not pose a risk that the consumer desires to hedge. In contrast, he does desire to hedge his demand shocks, but since θ_H has dropped out of his optimization problem, he only wishes to hedge the part of the demand shocks that governs his intertemporal choice—the state variable γ_H . The intuition for why θ_H drops out of the optimization is simple. The country wants to consume more of the Home good in the high- θ_H states, and so it desires higher wealth in those states. But these states are precisely the ones in which it desires fewer imports $(1 - \theta_H \text{ is low})$. The two effects exactly balance each other.

The next issue we need to address is market incompleteness. A technique for solving such problems in a single-good framework via martingale methods has been developed in a seminal contribution of He and Pearson (1991). He and Pearson show that, just like for the case of complete markets, one can replace the dynamic optimization problem (9)-(10) by the following static variational problem:

$$\max_{\mathbb{C}_H} E\left[\int_0^T e^{-\rho t} \gamma_H(t) \log \mathbb{C}_H(t) dt\right]$$
(11)

s.t.
$$E\left[\int_0^T \xi_{\nu}(t)\mathbb{C}_H(t)dt\right] \le W_H(0),$$
 (12)

where ξ_{ν} denotes an appropriate state price density—i.e., an Arrow-Debreu state price per unit of probability *P*. The difficulty arises from the fact that in incomplete markets, there is an infinite number of such state price densities consistent with no arbitrage and hence potentially an infinite number of static budget constraints (12). However, this set of budget constraints is known to possess some special structure. Let *m* denote the market price of risk process

$$m(t) \equiv \sigma(t)^{\top} (\sigma(t)\sigma(t)^{\top})^{-1} (\mu(t) - r(t)\mathbf{1}),$$
(13)

where $\mu \equiv (\mu_S, \mu_{S^*})^{\top}$ and **1** is a two-dimensional vector of ones. Then the set of state price densities can be represented as (He and Pearson, Proposition 1):

$$d\xi_{\nu}(t) = -r(t)\xi_{\nu}(t)dt - (m(t) + \nu(t))^{\top}\xi_{\nu}(t)d\vec{w}(t),$$
(14)

with $\nu(t) \in \mathbb{R}^4$ satisfying $\sigma(t)\nu(t) = 0$, $\forall t \in [0,T]$ and $\int_0^T ||\nu(t)||^2 dt < \infty$.⁶ It is easy to see that if the volatility matrix σ is a nondegenerate square matrix, the condition $\sigma(t)\nu(t) = 0$ can be satisfied only for $\nu(t) = \mathbf{0}$, where $\mathbf{0}$ is a four-dimensional vector of zeros. This is precisely the case when markets are complete: the state price density is unique and $\nu(t) = \mathbf{0}$ at all t. If, however, the volatility matrix has fewer rows than there are Brownian motions, many possible $\nu(t)$'s can satisfy the restriction $\sigma(t)\nu(t) = 0$. This is the case when markets are intrinsically incomplete.

He and Pearson go on to prove that there exists a unique individual-specific ν , which we denote by ν_H , that minimizes the maximum expected utility in (11). We derive the expression for it in the proposition below. The only relevant budget constraint in (12) is then the one corresponding to ν_H .

⁶The notation $||z||^2$ stands for the dot product $z \cdot z$.

Establishing the portfolio that solves the optimization problem (11)-(12) is then straightforward. We report this portfolio, as well as the portfolio of Foreign, in the following proposition.

Proposition 1. (i) The fractions of wealth x_H and x_F invested in the risky stocks by the Home and the Foreign country, respectively, are given by

$$x_{H}(t) = \underbrace{(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) m(t)}_{mean-variance \ portfolio} + \underbrace{(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) \frac{\sigma_{\gamma_{H}}(t)^{\top}}{\gamma_{H}(t)}}_{hedging \ portfolio}$$
(15)

$$x_F(t) = (\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) m(t).$$
(16)

The fractions of wealth invested in the bond by Home and Foreign are given by $1 - \mathbf{1}^{\top} x_H(t)$ and $1 - \mathbf{1}^{\top} x_F(t)$, respectively.

(ii) The processes ν_{H} and ν_{F} , entering the specification of the personalized state price densities of Home and Foreign, respectively, are given by

$$\nu_H(t)^{\top} = -\frac{\sigma_{\gamma_H}(t)}{\gamma_H(t)} (I_4 - \sigma(t)^{\top} (\sigma(t)\sigma(t)^{\top})^{-1} \sigma(t)) \quad and \quad \nu_F(t) = \mathbf{0},$$
(17)

where I_4 is a 4-dimensional identity matrix.

Consider first the portfolio of the Home consumer. It consists of two parts: the mean-variance efficient portfolio and the hedging portfolio. This decomposition is standard in the portfolio choice literature. The optimal mean-variance portfolio was first derived by Markowitz (1952) in a oneperiod setting and later generalized by Merton (1971) to a continuous-time stochastic environment. Furthermore, Merton shows that in addition to the mean-variance portfolio an investor optimally selects a hedging portfolio whose role is to offset fluctuations in the state variables in his optimization problem. As is well-known, investors with logarithmic preferences do not wish to hedge against changes in their investment opportunity set (stock and bond price dynamics)—in that sense they behave myopically. However, they do wish to hedge against fluctuations in the state variables entering their preferences, namely the preference shifts. When markets are complete (or effectively complete), the gains made by the hedging portfolio are perfectly positively correlated with the fluctuations in state variable Home desires to hedge: γ_H . (This is the state variable entering Home's objective function (9)). When markets are incomplete, not every payoff can be replicated and so it is typically not possible to construct a portfolio whose gains are perfectly correlated with a state variable. In that case, the Home investor chooses the portfolio most highly correlated with γ_H .

In contrast, the Foreign investor demands no hedging portfolio. This is because the term γ_F entering his objective function is non-stochastic. Consequently, the inability to hedge perfectly under incomplete markets does not hurt the Foreign investor: in contrast to that of the Home investor, his personalized ν_F remains the same as it would be under complete markets.

As we elaborate later, (heterogeneous) hedging demands is the key vehicle for generating trade in financial markets in equilibrium. For example, in the absence of preference shifts, agents have no hedging demands and hence they have no reason to trade assets.

2.3. Characterization of Equilibrium

The techniques used so far were developed for partial equilibrium analyses. We now embed these techniques in our general equilibrium framework. An equilibrium in our economy is defined in a standard way: it is a collection of goods and asset prices (p, p^*, S, S^*, B) and consumption-investment policies $(C_i(t), C_i^*(t), x_i^S(t), x_i^{S^*}(t)), i \in \{H, F\}$ such that (i) each consumer-investor maximizes his utility (7) subject to the budget constraint (6) and (ii) goods, stock, and bond markets clear.

In the economy with incomplete markets the equilibrium allocation would not be Pareto optimal. Hence, the usual construction of a representative agent's (planner's) utility as a weighted sum, with constant weights, of individual utility functions is not possible. Instead, we are going to employ a fictitious representative agent with stochastic weights (introduced in an important contribution by Cuoco and He (1994)), with these stochastic weights reflecting the effects of market incompleteness.⁷ This fictitious representative agent maximizes his utility subject to the resource constraints:

$$\max_{\{C_H, C_H^*, C_F, C_F^*\}} \qquad E\left[\int_0^T e^{-\rho t} \left(u_H(C_H(t), C_H^*(t)) + \lambda(t) \, u_F(C_F(t), C_F^*(t))\right) dt\right]$$
s. t.
$$C_H(t) + C_F(t) = Y(t) \,,$$
$$C_H^*(t) + C_F^*(t) = Y^*(t) \,,$$

where we have normalized the weight on the Home consumer to be equal to one and assigned the weight λ to the Foreign consumer. λ is simply the ratio of the marginal utilities of either good of the two countries. When markets are complete, this ratio is constant and the allocation is Pareto

⁷Alternatively, we could have solved for equilibrium directly from the system of equilibrium equations. We prefer the method we are presenting because of the clarity of the ensuing intuitions. The construction of a representative agent with stochastic weights has been employed extensively in dynamic asset pricing models with financial market frictions. See, for example, Basak and Croitoru (2000), Basak and Cuoco (1998), and Detemple and Serrat (2003). A related approach is the extra-state-variable methodology of Kehoe and Perri (2002) and Marcet and Marimon (1999). For the original solution method utilizing weights in the representative agent, see Negishi (1960).

optimal. When markets are incomplete, the marginal utilities are no longer proportional, and so λ becomes stochastic. It enters as an additional (endogenous) state variable, which reflects the relative importance the "planner" assigns to the Foreign country's utility.

Solving the representative agent's optimization problem, we obtain the sharing rules

$$C_H(t) = \frac{\gamma_H(t)\theta_H(t)}{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}Y(t), \quad C_H^*(t) = \frac{\gamma_H(t)(1-\theta_H(t))}{\gamma_H(t)(1-\theta_H(t)) + \lambda(t)\gamma_F\theta_F}Y^*(t),$$
(18)

$$C_F(t) = \frac{\lambda(t)\gamma_F(1-\theta_F)}{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}Y(t), \quad C_F^*(t) = \frac{\lambda(t)\gamma_F\theta_F}{\gamma_H(t)(1-\theta_H(t)) + \lambda(t)\gamma_F\theta_F}Y^*(t).$$
(19)

We can now derive the terms of trade that prevail in a competitive equilibrium. They are identified with the ratio of either country's marginal utilities of the Home and Foreign goods:

$$q(t) = \frac{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}{\gamma_H(t)(1-\theta_H(t)) + \lambda(t)\gamma_F\theta_F} \frac{Y^*(t)}{Y(t)}.$$
(20)

We next use the no-arbitrage valuation principle to obtain stock prices and equilibrium wealth of the countries.

Lemma 1. Equilibrium stock prices in our economy are given by

$$S(t) = \frac{1 - e^{-\rho(T-t)}}{\rho} \frac{q(t)}{aq(t) + 1 - a} Y(t), \qquad (21)$$

$$S^{*}(t) = \frac{1 - e^{-\rho(T-t)}}{\rho} \frac{1}{aq(t) + 1 - a} Y^{*}(t)$$
(22)

and the wealth of the countries by

$$W_H(t) = \frac{\gamma_H(t)}{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}S(t), \ W_F(t) = \frac{\lambda(t)\gamma_F}{\gamma_H(t)(1-\theta_H(t)) + \lambda(t)\gamma_F\theta_F}S^*(t).$$
(23)

Lemma 1 yields a simple interpretation of the weight λ . One can see that

$$\lambda(t) = \frac{W_F(t)}{W_H(t)} \frac{\gamma_H(t)}{\gamma_F}.$$
(24)

That is, incomplete markets enrich the dynamics of the economy with an additional state variable λ , which is related to the wealth distribution, but not given exactly by the wealth distribution unless $\gamma_H(t)$ is constant. We have already discussed the role of $\gamma_H(t)$ earlier in our analysis: it was the state variable giving rise to the hedging portfolio held by Home.

Lemma 1 allows us to characterize the dynamics of stock returns and the market price of risk in equilibrium, which are tedious but straightforward to compute. Equation (24) lets us pin down the weight λ . We relegate the details of the necessary calculations to the Appendix, and report the resulting dynamics of λ below.

Proposition 2. (i) In an equilibrium, the weight of the Foreign country in the fictitious representative agent follows

 $d\lambda(t) = -\lambda(t) \nu_H(t) d\vec{w}(t), \quad \text{with } \lambda(0) = \gamma_H(0)(1 - \theta_H(0)) / (\gamma_F(1 - \theta_F)).$

(ii) When such equilibrium exists, the volatility matrix σ , the market price of risk m, and hence ν_H can be computed as functions of exogenous state variables. They are reported in the Appendix.

Note that our characterizations the terms of trade, consumption, and stock prices presented in this section all involve the exogenous state variables of the model and one endogenous quantity: the weight λ . With this weight λ now characterized in Proposition 2, we can then pin down these equilibrium quantities and their dynamics. Moreover, the countries' portfolios held in equilibrium are also fully determined now, with the volatility matrix of stock returns and the market price of risk characterized fully in terms of exogenous state variables (see the Appendix). Admittedly, the equilibrium characterizations of the portfolios are not particularly transparent. To develop intuition, in Section 4 we consider several special cases in which the expressions for the portfolios are simple. The analysis of these special cases relies in part on the result of the following lemma.

Lemma 2. The countries hold no bond in their portfolios if and only if the value of the hedging portfolio demanded by Home is equal to zero.

Proof. Suppose that bondholdings of the countries are zero. This is equivalent to saying that the fraction of wealth each country invests in the stocks is equal to one:

$$\mathbf{1}^{\top} x_{H}(t) = \mathbf{1}^{\top} \left[(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) m(t) + (\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) \frac{\sigma_{\gamma_{H}}(t)}{\gamma_{H}(t)} \right] = 1$$

$$\mathbf{1}^{\top} x_{F}(t) = \mathbf{1}^{\top} (\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t) m(t) = 1,$$

where we have substituted the formulas for the portfolios derived in Proposition 1. This can happen only if $\mathbf{1}^{\top}(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t)\frac{\sigma_{\gamma_H}(t)}{\gamma_H(t)} = 0$ —i.e., the fraction of wealth invested in the hedging portfolio is zero.

Conversely, if the value of the hedging portfolio is zero, then $\mathbf{1}^{\top}x_{H}(t) = \mathbf{1}^{\top}x_{F}(t)$ = $\mathbf{1}^{\top}(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t)m(t)$. Bond market clearing then implies that $\mathbf{1}^{\top}(\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t)m(t) = 1$.

3. External Accounts

From the theoretical and practical point of view, the measure of external sustainability of countries has evolved through time. The original measure reflected simply the trade balance of goods, then it got revised to reflect the trade balance of goods and services, then switched to the (conventionally-defined) current account, and now it is changing again so as to better capture changes in net foreign asset positions. In policy circles, it is not uncommon to disregard all previous measures of external imbalances once a new measure comes to the fore.⁸

In this section, we define the trade balance, the conventional current account, and the capitalgains adjusted current account in the context of our model. The first one refers to the balance of imported and exported goods, the second one adds dividend and interest payments to the trade balance, and the last one adds the balance of capital gains to the conventional current account. Our goal is to study within our model the relationships between the traditional measures of sustainability, such as the trade balance and the current account, and the measure that has received most of the attention lately, the NFA position. We show that the trade balance and the capital-gains adjusted current account are closely intertwined, and link the conventional current account to the value of the countries' hedging portfolios.

3.1. The Trade Balance and the Conventional Current Account

In our model, the trade balance—defined as exports minus imports—is simply

$$TB_{H}(t) = (p(t)(Y(t) - C_{H}(t)) - p^{*}(t)C_{H}^{*}(t))dt$$

The conventional measure of the *current account* differs from the trade balance in that it also includes net dividend and interest payments (but not capital gains). For expositional simplicity, let us concentrate on the Home country. The conventionally-defined current account in our model is given by

$$CA_{H}(t) = TB_{H}(t) + \left[\underbrace{s_{H}^{S^{*}}(t)p^{*}(t)Y^{*}(t) - s_{F}^{S}(t)p(t)Y(t)}_{\text{Net Dividend Payments}} + \underbrace{s_{H}^{B}(t)B(t)r(t)}_{\text{Net Interest Payments}}\right]dt, \quad (25)$$

where s_i^j denotes the number of shares of asset j held by country i. The second and the third terms in (25) are dividend receipts from foreign assets minus dividend payments to Foreign, and the last

⁸As the former US Treasury Secretary Paul O'Neill once memorably remarked, the (conventional) current account had become a "meaningless concept."

term is the interest paid on current bondholdings. Recall that each of the above quantities in our model is defined as a rate (e.g., the export rate, the dividend rate, etc.) and hence need to be scaled by a time increment. This is the reason behind the term "dt" appearing in (25).

An often cited shortcoming of pure-exchange models with log-linear preferences is that they are unable to generate nontrivial current accounts. Having a current account equal to zero at all times would clearly hinder any quantitative analysis of current account deficits that we intend to undertake in this paper. It is therefore worth highlighting the situations under which the current account is zero in our model.

Lemma 3. The current account of the Home country can be represented as follows:

$$CA_{H}(t) = s_{H}^{B}(t)B(t)\left(r(t) - \frac{\rho}{1 - e^{-\rho(T-t)}}\right)dt.$$
(26)

Proof. Note that

$$\begin{split} s_{H}^{B}(t)B(t) &= W_{H}(t) - s_{H}^{S}(t)S(t) - s_{H}^{S^{*}}(t)S^{*}(t) \\ &= \frac{\gamma_{H}(t)}{\gamma_{H}(t)\theta_{H}(t) + \lambda(t)\gamma_{F}(1-\theta_{F})}S(t) - (1-s_{F}^{S}(t))S(t) - s_{H}^{S^{*}}(t)S^{*}(t) \\ &= \frac{1-e^{-\rho(T-t)}}{\rho} \left[p(t)Y(t)\frac{\gamma_{H}(t)(1-\theta_{H}(t)) - \lambda(t)\gamma_{F}(1-\theta_{F})}{\gamma_{H}(t)\theta_{H}(t) + \lambda(t)\gamma_{F}(1-\theta_{F})} - s_{H}^{S^{*}}(t)p^{*}(t)Y^{*}(t) + s_{F}^{S}(t)p(t)Y(t) \right], \end{split}$$

where the second equality follows from Lemma 1 and stock market clearing $(s_H^S(t) = 1 - s_F^S(t))$, and the last one, again, from Lemma 1. On the other hand, by substituting (18) into (25) and simplifying, one can show that

$$CA_{H}(t) = \left[-p(t)Y(t)\frac{\gamma_{H}(t)(1-\theta_{H}(t)) - \lambda(t)\gamma_{F}(1-\theta_{F})}{\gamma_{H}(t)\theta_{H}(t) + \lambda(t)\gamma_{F}(1-\theta_{F})} + s_{H}^{S^{*}}(t)p^{*}(t)Y^{*}(t) - s_{F}^{S}(t)p(t)Y(t) + s_{H}^{B}(t)B(t)r(t)\right]dt$$
$$= s_{H}^{B}(t)B(t)\left(-\frac{\rho}{1-e^{-\rho(T-t)}} + r(t)\right)dt.$$

This proves the statement in (26). \blacksquare

This lemma reveals that the first sufficient condition for the current account to be equal to zero is that the Home country (and hence the Foreign) holds no bonds. While it is indeed a common implication of models with log-linear preferences to have zero net bond holdings in equilibrium, nothing in our model prevents the bond holdings from being different from zero. That is, preference shifts may potentially introduce enough heterogeneity among the countries so that they are willing to trade in all available financial assets for risk-sharing purposes. The second condition under which the current account is zero is when the interest rate r(t) is equal to $\rho/(1 - e^{-\rho(T-t)})$. The latter quantity is deterministic, while the interest rate is a stochastic process. Hence, it is true only on the measure zero set of parameter values.

3.2. The Capital-Gains Adjusted Current Account

We now define the current account as the change in the net foreign asset position of a country. Unlike the conventional current account CA, this measure fully accounts for the capital gains on the NFA positions. We therefore label it "CGCA," which stands for the "capital-gains adjusted current account."

$$CGCA_{H}(t) \equiv dNFA_{H}(t) = d\left[s_{H}^{S^{*}}(t)S^{*}(t) - s_{F}^{S}(t)S(t) + s_{H}^{B}(t)B(t)\right],$$
(27)

where the first two terms in the square brackets are Home's investment in the Foreign stock minus Foreign' investment in the Home stock, and the last term is Home's balance on the bond account.

Note that, by market clearing, $s_F^S(t) = 1 - s_H^S(t)$ and that, by definition, Home's financial wealth equals its portfolio value, $W_H(t) = s_H^S(t)S(t) + s_H^{S^*}(t)S^*(t) + s_H^B(t)B(t)$. Hence, we can rewrite (27) as

$$CGCA_H(t) = dW_H(t) - dS(t).$$
⁽²⁸⁾

3.3. Congruence between NFA and Trade Balance

To conclude this section, we derive the expression for NFA in our model and draw a connection between the NFA position and the trade balance. Note that

$$NFA_{H}(t) = W_{H}(t) - S(t)$$

$$= \frac{1}{\xi_{\nu_{H}}(t)} E_{t} \left[\int_{t}^{T} \xi_{\nu_{H}}(s) \left(p(s) C(s) + p^{*}(s) C^{*}(s) \right) ds \right] - \frac{1}{\xi_{\nu_{H}}(t)} E_{t} \left[\int_{t}^{T} \xi_{\nu_{H}}(s) p(s) Y(s) ds \right]$$

Hence, by definition of a trade balance,

$$NFA_{H}(t) = -\frac{1}{\xi_{\nu_{H}}(t)}E_{t}\left[\int_{t}^{T}\xi_{\nu_{H}}(s)\underbrace{(p(s)(Y(s) - C_{H}(s)) - p^{*}(s)C_{H}^{*}(s))ds}_{=TB_{H}(s)}\right].$$
(29)

Equation (29) is nothing else but the familiar statement that the NFA position is given by the present value of the future trade deficits. The traditional intertemporal approach to external

adjustment—that ignores changes in the state price density (or the stochastic discount factor) ξ_{ν_H} —says that, for example, for a country with a negative NFA position, adjustment must come through future trade surpluses. Recent literature challenges this conclusion and draws attention to the "valuation channel" of the external adjustment that operates precisely through changes in the stochastic discount factor (Gourinchas and Rey (2007b)). It is argued that such changes are large and volatile, and hence the valuation channel should have a substantial contribution to the NFA dynamics. Surprisingly, it turns out that in our model, after the endogenous responses of asset prices and hence the stochastic discount factor to underlying shocks are taken into account, the NFA adjustment takes place instantaneously and entirely through the trade balance. In that sense, the traditional and the new views are not at all inconsistent.⁹

Lemma 4 (Congruence between NFA and trade balance). The relationship between the net foreign assets and the trade balance is given by

$$TB_{H}(t) = -\frac{\rho}{1 - e^{-\rho(T-t)}} NFA_{H}(t)dt.$$
(30)

The net foreign asset position of Home is

$$NFA_H(t) = \frac{\gamma_H(t)(1-\theta_H(t)) - \lambda(t)\gamma_F(1-\theta_F)}{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}S(t).$$
(31)

We note that this result is not due to the fact that we only consider permanent shocks. For example, the supply shocks in our model are general stochastic processes and in particular, they can be mean-reverting. In the related literature, a similar result has been obtained as a first-order approximation (Tille and van Wincoop (2010)). Here, the formula is exact and approximation-free. The perfect negative correlation between the current trade balance and the NFA position certainly relies on our assumption that the agents have log-linear preferences that rule out intertemporal hedging motives. It is important to evaluate the robustness of this result under alternative preferences that give rise to intertemporal hedging.

Our congruence relationship may remind the reader of the textbook formula from the intertemporal approach to the current account (Obstfeld and Rogoff (1996), Chapter 2). That formula is derived in the context of an economy with no stocks, under many simplifying assumptions. The economy we are considering is significantly more complicated—featuring endogenous portfolio recompositions and trade in equities under incomplete markets—and yet the relationship between the

⁹It is important to note that the "connection" is between the NFA and the trade balance of goods and services, and not between NFA and the conventional current account.

NFA position and the trade balance comes out similar to that in the classical external adjustment theory.

4. Equilibrium Portfolios

As we have been stressing, in our model there is a close relationship between the countries' external accounts and their equilibrium portfolio compositions. These interconnections take place through different mechanisms—related to the terms of trade, asset prices, hedging motives, or portfolio choices. In this section, we present special cases of our model that highlight such mechanisms. To establish a benchmark, we start by replicating some results of the earlier literature—in particular, that under certain conditions financial markets could be irrelevant for the allocation of consumption. In this special case of our model, there is perfect risk sharing and the stock market returns are perfectly correlated around the world. Risk diversification occurs through terms of trade fluctuations. Second, we study the case in which the stock market returns are not perfectly correlated, but the optimal portfolios again fully diversify international risk. In this case, the channel of diversification comes from the countries' holdings, and not (entirely) from the terms of trade. The case we study is one in which shocks to any of the countries have no implications for the conventional current account, portfolio holdings are constant through time and therefore. capital flows are zero. The third example is one in which markets are intrinsically incomplete and hence risk sharing is imperfect. We have set up this example in such a way that the value of the hedging portfolio is zero, which in turn implies that the bond holdings and the conventional current account are always equal to zero. However, the capital-gains adjusted current account does vary in response the underlying shocks because asset prices and optimal portfolios are not constant. On a separate note, it is important to stress that even though in this example markets are incomplete, the (conventional) current account is always zero. Finally, we study a case in which the value of the hedging portfolio is different from zero. We specialize our setup so that markets are complete, but all external accounts are nonzero. We analyze the relationship between the current account and portfolio compositions in this environment. In summary, the objective of this section is to cumulatively develop the intuitions behind the interconnections that our model exhibits, which we do via examples that capture several different aspects of the workings of the model.

4.1. Example 1: The Irrelevance Result

The first example we study is one in which the financial markets' structure is irrelevant, there are no net portfolio flows, and therefore, capital gains on financial assets play no role in the international adjustment process. This is the case considered in Helpman and Razin (1978), Cole and Obstfeld (1991), and Zapatero (1995). The three examples that follow relax some of the assumptions we make here in order to clarify the role that capital gains play.

In our model, we obtain our irrelevance result by specializing the Home consumer's preferences so that γ_H and θ_H are constant. As is well-known, under this specification the returns on the two stocks are perfectly correlated in equilibrium. Hence, portfolio allocations into these stocks are indeterminate. Only positions in the composite stock market, $S(t) + S^*(t)$, can be uniquely determined. Using an argument analogous to that we employed in Section 2.2, we can derive the investors' optimal fractions of wealth invested in the composite stock market. One can easily see that the portfolios demanded by the two agents are going to be identical: in the absence of preference shifts, they both demand the mean-variance but no hedging portfolios.

It is also well-known that financial markets are effectively complete in this special case. This is equivalent to saying that $\nu_H(t) = \nu_F(t) = 0$ at all times and hence, from Proposition 2, the weight λ is constant. Thus, Pareto optimality obtains despite market incompleteness—or, in other words, markets are effectively complete. Investors are not adversely affected by market incompleteness because they do not make use of financial markets to construct portfolios hedging against fluctuations in any state variables: there is no state variable either agent desires to hold a hedge against. The intuition for why the financial markets are not needed in this case comes from the fact that movements in the terms of trade exactly offset output shocks and hence the values of the dividends on the Home and the Foreign stock markets, p(t)Y(t) and $p^*(t)Y^*(t)$, respectively, are always the same (up to a multiplicative constant). Fluctuations in the terms of trade therefore fully offset the supply shocks; i.e., with no demand uncertainty the capital gains on the two stocks are always perfectly correlated. This feature of our model is due to the way we specified preferences (log-linear) and endowments (shares of trees), and represents a simple benchmark for comparison.

Finally, unlike holdings of individual stocks, the bond holdings of the countries are uniquely determined: there are equal to zero at all times.¹⁰ This is because, the two countries demand the

¹⁰This result is not new. See Cass and Pavlova (2004).

same portfolio and in particular, wish to invest the same fraction of wealth in the bond. For the bond market to clear, this fraction has to be zero. Consequently, CA(t) = 0 at all times. Moreover, the capital-gains adjusted current account CGCA is also zero. Each country is holding the same portfolio and stock markets are perfectly correlated. Therefore, the NFA positions (and changes in the NFA positions) are zero at all times.

4.2. Example 2: No Hedging and No Capital Flows

The purpose of this example is to highlight the importance of international asset cross-holdings and portfolio rebalancing. We consider a version of our model in which portfolio holdings are unique (as opposed to indeterminate as in the previous case), but portfolio holdings are fixed (hence there is never a rebalancing after any of the shocks). We show that in this case the current account is zero at all times. Any movements in the capital-gains adjusted current account are due to fluctuations in asset prices, and not due to portfolio rebalancing.

To highlight these dynamics, consider the special case of the model in which γ_H remains constant $(\sigma_{\gamma_H}(t) = \mathbf{0})$, and θ_H is stochastic. In the presence of preference shocks—even one type of shock, as we specify here—the two stocks are no longer perfectly correlated, the volatility matrix is invertible, and hence the expressions in Propositions 1 and 2 readily apply. It turns out that the equity portfolios of the countries, expressed as numbers of shares, take a particularly simple form

$$s_{H}(t) = \left(\frac{\gamma_{H}}{\gamma_{H} + \lambda(t)\gamma_{F}}, \frac{\gamma_{H}}{\gamma_{H} + \lambda(t)\gamma_{F}}\right) \quad \text{and} \quad s_{F}(t) = \left(\frac{\lambda(t)\gamma_{F}}{\gamma_{H} + \lambda(t)\gamma_{F}}, \frac{\lambda(t)\gamma_{F}}{\gamma_{H} + \lambda(t)\gamma_{F}}\right), \quad (32)$$

where $s_i \equiv (s_i^S, s_i^{S^*})$ are obtained from x_i using Lemma 1.

No hedging portfolio is held by either country and $\nu_H(t) = 0$. These results is immediate from Proposition 1. Intuitively, the only state variable that the Home investor desires to hedge against in our economy is γ_H , and that variable is non-stochastic. It is of no surprise then that it turns out that $\nu_H(t) = 0$: the investor is able to achieve the same efficiency of hedging as under complete markets. Consequently (Proposition 2), again, the weight λ is constant and markets are effectively complete. Since λ is constant, the portfolios reported above are also constant, and hence there are no capital flows between countries at any time. In contrast to the no preference shifts case, however, one can see that effective market completeness does not lead to the indeterminacy of equilibrium portfolios.

This example is also useful in drawing a distinction between portfolio home bias and consump-

tion home bias. In our model, the parameters controlling the degree of home bias are θ_H and θ_F . In other words, the home bias is coming because the demand shock is affecting home demand for home goods (this is equivalent to explicitly modeling shocks to the non-tradable demand which has been already highlighted in the literature). It is important to highlight that the home bias in consumption has *nothing* to do with the home bias in portfolios in this case. As one can see from (32), the countries always hold an identical number of shares of the Home and Foreign stock.

Furthermore, just like in the previous special case, the countries invest nothing in the bond. To see this, recall that the hedging portfolio is zero at all times. The result then follows from Lemma 2. Intuitively, there is not enough heterogeneity across countries to justify borrowing and lending. With γ_H being constant, the Home agent has no desire to hedge his demand shocks. Consequently, the two countries demand the same portfolio (in fractions of wealth; see Proposition 1) and in particular, wish to invest the same fraction of wealth in the bond. For the bond market to clear, this fraction has to be zero.

Finally, note that

$$CA_i(t) = 0, \qquad i \in \{H, F\}.$$

The (conventional) current account is zero because none of the countries invests in the bond (Lemma 3). Except for time t = 0, the trade balance and the capital-gains adjusted current account are generally non-zero, however. Although there is no portfolio rebalancing (capital flows) taking place, the capital-gains adjusted current account does fluctuate. This is because the prices of stocks move in response to the underlying shocks. These stock price fluctuations result in capital gains that get transmitted to the capital-gains adjusted current account. We examine the effects of the shocks on the capital-gains adjusted current account in the next section.

4.3. Example 3: Incomplete Markets

We now consider a more general case of our model in which the current account is still identically equal to zero but the hedging portfolio is now different from zero and there are capital flows across countries. The external adjustment process in this example is therefore driven entirely by the valuation effects and has nothing to do with the traditional channels.

We consider a special case of the model in which the demand shocks γ_H and θ_H load only on the Brownian motions w^{γ} and w^{θ} , and not on w and w^* —i.e., the demand shocks are completely independent of the supply shocks. Formally, we assume that $\sigma_{\gamma_H}(t) = (0, 0, \sigma_{\gamma_1}(t), \sigma_{\gamma_2}(t))$ and $\sigma_{\theta_H}(t) = (0, 0, \sigma_{\theta_1}(t), \sigma_{\theta_2}(t))$, with $\sigma_{\gamma_H} > 0$ and $\sigma_{\theta_H} > 0$. We derive the stockholdings of Home and Foreign, respectively, to be as follows:

$$s_{H} = \frac{1}{G} \left(\lambda \gamma_{F} \theta_{F} \sigma_{\gamma_{H}} \sigma_{\theta_{H}}^{\top} + \gamma_{H}^{2} ||\sigma_{\theta_{H}}||^{2}, \quad -\lambda \gamma_{F} (1 - \theta_{F}) \sigma_{\gamma_{H}} \sigma_{\theta_{H}}^{\top} + \gamma_{H}^{2} ||\sigma_{\theta_{H}}||^{2} \right), \tag{33}$$

$$s_F = 1 - s_H \,, \tag{34}$$

where $G \equiv \gamma_H^2 ||\sigma_\theta||^2 (1 + \lambda \gamma_F) - \lambda \gamma_F (1 - \theta_H - \theta_F) \sigma_{\gamma_H} \sigma_{\theta_H}^{\top}$. In the expressions above and for the remainder of this section, we suppress the argument t. It already becomes clear at this point that depending on parameter values, our model can produce large gross portfolios.

The hedging portfolio, h, held by Home, in numbers of shares, is

$$h = \frac{1}{G} \left((\lambda \gamma_F \theta_F + \gamma_H (1 - \theta_H)) \sigma_{\gamma_H} \sigma_{\theta_H}^{\top}, \ (\lambda \gamma_F (1 - \theta_F) + \gamma_H \theta_H) \sigma_{\gamma_H} \sigma_{\theta_H}^{\top} \right) . \tag{35}$$

Consider the gain on the hedging portfolio, which turns out to be:

$$dh = [\ldots] dt + \frac{W_H \sigma_{\gamma_H} \sigma_{\theta_H}^\top}{\gamma_H ||\sigma_{\theta_H}||^2} (0, 0, \sigma_{\theta_1}, \sigma_{\theta_2}) d\vec{w},$$
(36)

and compare it to the fluctuations in the state variable, γ_{H} , that Home desires to hedge against:

$$d\gamma_H = (0, \ 0, \ \sigma_{\gamma_1}, \ \sigma_{\gamma_2}) \, dec{w}_{\gamma_1}$$

(Recall that the discount factor γ_H is the state variable entering Home's objective function (9).) Comparing the above two equations, one can see that the hedge is generally not perfect. The extent to which a country is able to hedge its γ_H shocks depends on the correlation between γ_H and θ_H (or on their covariance $\sigma_{\gamma_H}\sigma_{\theta_H}^{\top}$). One extreme is when they are perfectly correlated, in which case the hedge is perfect and hence markets become effectively complete. In that case, the dynamics of *h* matches the dynamics of γ_H (the vectors $(0, 0, \sigma_{\theta_1}, \sigma_{\theta_2})$ and $(0, 0, \sigma_{\gamma_1}, \sigma_{\gamma_2})$ are collinear). The other extreme is when the correlation is zero (and, of course, $\sigma_{\gamma_H}\sigma_{\theta_H}^{\top} = 0$), in which case the hedging portfolio is constant and agents are unable to hedge any of the γ_H -risk.

Note that the holdings of the two stocks in the hedging portfolio have the opposite sign, and this sign depends on the sign of G. This implies that our model can produce a home bias or a reverse home bias in portfolios. Note that in this case the presence of home bias in consumption $(\theta_H > 1/2 \text{ and } \theta_F > 1/2)$ is sufficient to generate home bias in portfolios.

The inability to hedge perfectly is indicative of the fact that market incompleteness matters. Indeed, in equilibrium,

$$\nu_{H} = \left(0, \quad 0, \quad \frac{\sigma_{\theta_{2}}(\sigma_{\theta_{1}}\sigma_{\gamma_{2}} - \sigma_{\theta_{2}}\sigma_{\gamma_{1}})}{\gamma_{H}||\sigma_{\theta}||^{2}}, \quad -\frac{\sigma_{\theta_{1}}(\sigma_{\theta_{1}}\sigma_{\gamma_{2}} - \sigma_{\theta_{2}}\sigma_{\gamma_{1}})}{\gamma_{H}||\sigma_{\theta}||^{2}}\right), \tag{37}$$

and hence the weight λ follows a stochastic process. The zeros in the first two positions of ν_H are not accidental. Since the preference shifts that the Home country faces are uncorrelated with the output shocks, it demands a hedge correlated with the Brownian motions w^{γ} and w^{θ} but not with w and w^* . Constructing such a hedging portfolio is possible: one can easily show that any zerocost portfolio of the two stocks is going to be uncorrelated with the output shocks. The hedging portfolio h must then have a value of zero, and one can easily verify from (35) and Lemma 1 that this is indeed the case. As a corollary, none of the countries holds any bond (Lemma 2) and hence their (conventional) current accounts are zero (Lemma 3):

$$CA_i(t) = 0, \qquad i \in \{H, F\}.$$

In this example, however, the capital-gains adjusted current account is different from zero. We examine it closely in the next section.

Remark 1 (A limiting case of "small risks"). There are two types of risks that we highlight in this example: the demand shocks due to fluctuations in γ_H and those due to the fluctuations in θ_H . The limit of $\sigma \gamma_H \to 0$ is discussed in detail in the previous subsection. The main conclusion that becomes apparent from comparing the case of stochastic and deterministic γ_H is that if uncertainty about γ_H disappears, so does market incompleteness.

Let us now examine the case of uncertainty about θ_H becoming small. We first establish whether markets remain incomplete. To do so, we take the limit in (37) as $\sigma_{\theta_1} \to 0$ and $\sigma_{\theta_2} \to 0$. One can easily see that this limit does not exist. For example, if we first let $\sigma_{\theta_1} \to 0$ and then $\sigma_{\theta_2} \to 0$, the limit of ν_H is $(0, 0, -\sigma_{\gamma_1}/\gamma_H, 0)$, and if we do the reverse, it is $(0, 0, 0, -\sigma_{\gamma_2}/\gamma_H)$. Hence, the limit does not exist. This is alarming because many of the existing methods for solving portfolio models in international macro-finance rely on approximations around a deterministic case. The limit of the countries' portfolios, however, is well-defined. Home's holdings of the two stocks converge to

$$s_{\scriptscriptstyle H} = \left(rac{ heta_{\scriptscriptstyle F}}{ heta_{\scriptscriptstyle H} + heta_{\scriptscriptstyle F} - 1}, -rac{1 - heta_{\scriptscriptstyle F}}{ heta_{\scriptscriptstyle F} + heta_{\scriptscriptstyle H} - 1}
ight) \,.$$

If we complete the markets in our model by adding two more bonds—such as, for example, the local-good denominated bonds—we are going to get the same portfolios as the ones occurring in this limit.

4.4. Example 4: Complete Markets and the Current Account

In the examples we have considered so far, we have obtained rich implications for asset prices, portfolio holdings and therefore on the NFA positions (which we discuss further in the next section). All three examples, however, imply that the conventional current account is zero at all times. The main reason is that in none of the examples agents hold any bond positions. In this section, we consider a special case of our model in which bondholdings are nontrivial. For simplicity, we set up the example in such a way that markets end up being complete. This highlights that market incompleteness is neither necessary nor sufficient to generate movements in the (conventional) current account.

To induce the countries to trade in bonds, we now allow for the correlation between the demand and output shocks. We reduce the number of Brownian motions driving the economy from four to two. In particular, we shut down Brownian motions w^{γ} and w^{θ} and require that all processes are adapted to the filtration generated by the output shocks w and w^* . Under this modification, all four-dimensional vectors in our analysis in Section 2 and the Appendix become two-dimensional. This implies further that the volatility matrix of stock returns σ is a 2 × 2 square matrix. If this matrix is nondegenerate—which is always the case in the presence of stochastic preference shifts financial markets are complete. Equilibrium allocation is then Pareto optimal and the weight λ is constant. It is important to emphasize that a constant λ does not imply constant wealth distribution; as one can see from equation (24), the relative wealth fluctuates together with γ_H .

In the interest of space, we do not report the countries' portfolios in this case. It suffices to say that now portfolios depend on all of the parameters of the model except for the drifts of outputs. As to be expected, the gain on the hedging portfolio in this case is perfectly correlated with the fluctuations in γ_H :

$$dh = [\ldots] dt + \frac{W_H}{\gamma_H}(\sigma_{\gamma_1}, \sigma_{\gamma_2}) d\vec{w},$$

which simply reflects the fact that when markets are complete, agents can perfectly hedge against fluctuations in pertinent state variables. In contrast to all the special cases we have considered so far, however, the value of the hedging portfolio is not equal to zero. Lemma 2 then implies that now the countries engage in borrowing and lending. Let us consider an example in which we set $\sigma_{\gamma_H} = (\sigma_{\gamma_1}, 0)$ and $\sigma_{\theta_H} = (0, \sigma_{\theta_2})$. In this example, the value of the bondholdings of the Home country becomes

$$-\frac{\frac{\left(1-e^{-\rho(T-t)}\right)}{\rho}YY^*\lambda\gamma_H\gamma_F\sigma_{\gamma_1}\sigma_{\theta_2}}{(1-a)Y\lambda\gamma_F(\gamma_H(1-\theta_H)+\lambda\gamma_F\theta_F)(1-\theta_H-\theta_F)\sigma_{Y^*}\sigma_{\gamma_1}+aY^*\gamma_H(\gamma_H+\lambda\gamma_F)(\gamma_H\theta_H+\lambda\gamma_F(1-\theta_F))\sigma_Y\sigma_{\theta_2}}$$

and under some restrictions on parameters, we can get this quantity to be always below zero. This implies that Home borrows from Foreign to finance its hedging portfolio, whose value is always greater than zero in this case. This example demonstrates that in our model it is possible to have a negative bond position and hence possibly a current account deficit (Lemma 3) forever. This does not in any way contradict sustainability of a country's external position: if an equilibrium exists, the budget constraints of both countries are always satisfied, and so a negative position in the bond account is offset by positive positions in the stocks. Having a negative debt position in equilibrium implies that Home is holding a leveraged portfolio. That portfolio is riskier than that of the Foreign country. The Home country is compensated for taking a greater share of world risk by collecting risk premia on its risky asset holdings. These risk premia can be used to finance current account deficits.¹¹ In a recent paper Gourinchas and Rey (2007a) argue that the US, whose external liabilities are mainly Treasury bills and bonds (with relatively low expected returns) while its external assets are riskier instruments (with relatively high expected returns), collects additional income from positive risk premia on its NFA positions. Our model allows for a calibration that captures this possibility.

According to Lemma 3, the countries' current accounts are always nonzero. This is the first time we encounter a nonzero current account in this section. As the case we are considering here demonstrates, enough heterogeneity in hedging demands that is sufficient to give rise to trade in bonds for risk sharing purposes guarantees that the current account deviates from zero.

5. Capital Gains and External Adjustment

We now explore the dynamics of NFA in our model, and, in particular, how NFA respond to demand and supply shocks. Let us focus on the NFA of, say, Home. To understand the intuition behind the role the capital gains in the dynamics of NFA, note first that the budget constraint of Home (6) can be equivalently represented as

$$dW_{H}(t) = \left[s_{H}^{B}(t) B(t) r(t) + s_{H}^{S}(t) S(t) \mu_{S}(t) + s_{H}^{S^{*}}(t) S^{*}(t) \mu_{S^{*}}(t)\right] dt \\ + \left[s_{H}^{S}(t) S(t) \sigma_{S}(t) + s_{H}^{S^{*}}(t) S^{*}(t) \sigma_{S^{*}}(t)\right] d\vec{w}(t) + TB_{H}(t) - p(t)Y(t) dt.$$

We can then substitute this expression into (28) and use (4) and the stock market clearing

$$CGCA_{H}(t) = TB_{H}(t) + \left[s_{H}^{S^{*}}(t)S^{*}(t)\mu_{S^{*}}(t) - s_{F}^{S}(t)S(t)\mu_{S}(t) + s_{H}^{B}(t)B(t)r(t)\right]dt + \left[s_{H}^{S^{*}}(t)S^{*}(t)\sigma_{S^{*}}(t) - s_{F}^{S}(t)S(t)\sigma_{S}(t)\right]d\vec{w}(t),$$
(38)

Define an *unexpected* change in a variable Z at time t as the diffusion term, $\sigma_Z(t)d\vec{w}$, in the dynamics $dZ(t) = \mu_Z(t)dt + \sigma_Z(t)d\vec{w}(t)$. (The trend term, $\mu_Z(t)dt$, is the *expected* change.) The un-

 $^{^{11}\}mathrm{We}$ thank Mick Devereux for suggesting this intuition to us.

expected changes dominate the short-term fluctuations in the NFA positions because the Brownianmotion term dwarfs the trend term. We therefore focus on the unexpected capital gains. They represent the movement in the NFA in response to a shock, on impact. Making use of (4)–(5) and (25), we can rewrite (38) as

$$CGCA_{H}(t) = CA_{H}(t) + s_{H}^{S^{*}}(t)dS^{*}(t) - s_{F}^{S}(t)dS(t)$$
(39)

From (39), we can see that, on impact, only valuation changes of the two stocks contribute to the unexpected capital gains on the NFA. There are no unexpected capital gains on the bond, since its return over the next instant is riskless.¹² The conventional current account is also riskless over the next instant.

We find that in our model the signs of the responses of the NFA to the underlying innovations depend on whether the country is a net debtor (NFA < 0) or a net creditor (NFA > 0). This generalizes an important result derived by Kraay and Ventura (2000). Recall from Lemma 4 that

$$NFA_{H} = \frac{\gamma_{H}(1-\theta_{H}) - \lambda \gamma_{F}(1-\theta_{F})}{\gamma_{H}\theta_{H} + \lambda \gamma_{F}(1-\theta_{F})} S.$$

We thus need the following condition:

Condition NC: Home is Net Creditor. $\gamma_H(1-\theta_H) - \lambda \gamma_F(1-\theta_F) > 0.$

To keep our discussion focused, we now specialize our economy to the setting in Example 3. In order to better highlight the effects of incomplete markets, we postulate further that the demand shock γ_H loads only on the Brownian motion w^{γ} and the demand shock θ_H only on the Brownian motion w^{θ} . That is, the demand shocks are not only independent of the output shocks but also independent of each other. Formally, we assume that $\sigma_{\gamma_H}(t) = (0, 0, \sigma_{\gamma_1}(t), 0)$ and $\sigma_{\theta_H}(t) =$ $(0, 0, 0, \sigma_{\theta_2}(t))$, with $\sigma_{\gamma_1} > 0$ and $\sigma_{\theta_2} > 0$. A similar analysis can be performed in the context of Example 2, but we focus on Example 3 for more generality. Our goal is to sign (or characterize) the direction of the valuation effects. Table 1 presents the unexpected gains/losses on the terms of trade, the stocks and the ensuing gains/losses on the NFA position of the Home country.

Table 1 reveals that on impact, both stocks yield unexpected capital gains in response to a positive output shock in either country $(dw \text{ or } dw^*)$.¹³ This is because a positive output shock in say, Home, raises the dividend on the Home tree. In response, the Home stock goes up. At the

¹²An important and realistic extension would be to introduce long-maturity bonds. Capital gains on these bonds will be part of the (unexpected) capital gains on NFA. We leave this extension for future research.

¹³To establish this result, we have explicitly computed the diffusion terms $\sigma_S S$ and $\sigma_{S^*} S^*$ in the equilibrium processes for the stock prices (4)–(5) and signed them. See the proof of Proposition 2 in the Appendix.

	dw	dw^*	dw^{γ}	dw^{θ}
Unexpected change in q	_	+	0	+
Unexpected change in S	+	+	0	+
Unexpected change in S^*	+	+	0	_
Unexpected change in NFA_H	$+^{NC}$	$+^{NC}$	0	_

Table 1: The valuation effects: Unexpected gains on the terms of trade, stocks and Home's the net foreign assets in response to the underlying shocks. The superscript NC indicates that a necessary and sufficient condition under which a sign obtains is Condition NC.

same time, it causes a deterioration of Home's terms of trade because the Home good becomes less scarce. This in turn improves Foreign's terms of trade and hence raises the value of the output of the Foreign tree. Hence, the Foreign stock rises. An analogous argument applies for the output shock at Foreign. Output shocks, then, always cause the two stock markets to comove (as in Cole and Obstfeld (1991)). The reaction of stock prices to the demand shocks has a distinctly different pattern: stock prices always move in opposite directions. As Home shifts its preference towards the Home good (in response to a positive realization of dw^{θ}), there is an excess demand for the Home good in the world. This pushes the price of the Home good up, or equivalently, causes an appreciation of the terms of trade, q. This raises the value of the Home output relative to Foreign. Consequently, the price of the Home stock increases, while that of the Foreign stock falls. An analogous argument applies for the demand shift towards the Foreign good (a negative realization of dw^{θ}).

The most subtle of the impact responses presented in Table 1 is that to a shock to the discount factor γ_H . At the outset, one would expect that such a shock should have an impact on the terms of trade and stock prices because Home desires to consume more at times when γ_H is high. For example, if it prefers one good over the other (e.g., if it has consumption home bias), then, following a shock to γ_H , the price of that good should go up, in relative terms. This intuition would have been correct under complete markets, and the fact that it fails here highlights the effects of market incompleteness in our model. Indeed, the Home agent would have liked to consume more in high- γ_H states, but we have set up our example in such a way that he is unable to purchase a state-contingent claim that allows him to do so. Recall from our discussion of equation (36) that the Home agent's ability to hedge γ_H shocks is governed by the correlation between θ_H and γ_H . When the correlation is perfect, the agent is able to hedge perfectly, and markets are complete. The opposite case—the case of zero correlation that we highlight in this example—features an extreme degree of market incompleteness, in which no hedging of γ_H is possible. Because of this, neither consumption nor prices respond to w^{γ} shocks.

To determine the unexpected capital gains on the NFA following a shock, we use equation (39). All we need to know is the capital gains on Home and Foreign stock and the countries' portfolio holdings. In the case we are considering here, the countries' portfolios take an especially simple form. Equations (33)–(34) reduce to

$$s_H = \left(\frac{\gamma_H}{\gamma_H + \lambda\gamma_F}, \frac{\gamma_H}{\gamma_H + \lambda\gamma_F}\right), \quad s_F = \left(\frac{\lambda\gamma_F}{\gamma_H + \lambda\gamma_F}, \frac{\lambda\gamma_F}{\gamma_H + \lambda\gamma_F}\right).$$

Importantly for our discussion here, both countries have positive holdings of each stock.

We are now ready to examine the adjustment of the NFA following a shock. The mechanism is the simplest for the demand shocks. Following a discount-factor shock (a movement in w^{γ}), obviously, there is no response of the NFA because the stock prices remain the same. Now consider a preference shift towards the Home good (a positive dw^{θ}). Home stock rises and foreign residents make gains on their holdings at Home, while Foreign stock falls and home residents suffer capital losses on their position abroad. Hence, the NFA of Home falls. A preference shift towards the Foreign good (a negative dw^{θ}) obviously causes the opposite reaction. Home stock falls and Foreign stock rises. Hence, Home gains on its investment in the Foreign stock while Foreign loses on its investment in the Home stock, and therefore the NFA of Home increases.

The responses to the output shocks are somewhat more difficult to analyze. This is because output shocks induce comovement of stocks across countries. A positive output shock makes both the Home and the Foreign stock go up. So both countries make a gain on their positions abroad simultaneously. The NFA movement then must reflect the relative magnitude of the gains to Home vs. Foreign investors, which in turn depends on their respective stock positions abroad. Let us consider two extremes. Suppose first that Home holds most of the supply of the risky stocks. Because none of the countries holds the bond, Home is holding most of the world's assets, which makes it a net creditor to the world. In response to a positive output shock, Home enjoys significant capital gains, both at home and abroad. The capital gains of Foreign are significantly smaller (in both markets) because Foreign's stock positions are much smaller. Hence, the net capital gain of Home a is larger and therefore Home's NFA increase. The other extreme is when Foreign holds most of the supply of the risky stocks, and hence is a net creditor. Then, the capital gains on Foreign's stock positions (both at home and abroad) by far exceed those of Home. Hence, Foreign's NFA increase, which implies that Home's NFA position deteriorates. The knife-edge case turns out to be the one in which Home turns from being a net creditor to a net debtor to the world, which is reflected in the signs reported in Table 1. The responses to an output shock at Foreign follow the same pattern.

It is important to emphasize that in our model the response of a country's NFA to an output shock is positive if and only if the country is a net creditor (i.e., under Condition NC). So only for countries like China or Japan, who are net creditors to the world, do the NFA positions improve following good news about productivity. But for a country like the US or the UK, whose NFA are large and negative, the NFA position should deteriorate in response to good news about productivity. The opposite, of course, is true for a drop in output (a recession) in either country.

6. Discussion

Some of the recent literature has drawn attention to the relevance of the quality of international assets for the discussion of global imbalances (see Caballero, Farhi, and Gourinchas (2008), Dooley, Folkerts-Landau, and Garber (2004), and Blanchard, Giavazzi, and Sa (2005) for the link to global imbalances, and Kouri (1982) for an earlier discussion). Because in our model the entire output of each country is capitalizable and there are no restrictions on capital flows, financial assets do not vary in their quality. But we believe that differences in asset quality is an important feature of international capital markets, and therefore it would be interesting to extend our framework to include this element into the analysis.

Extending the framework beyond log-linear preferences may also prove fruitful. This would introduce some of the intertemporal hedging motives that have been shut down in our model. Moving away from the log-linear specification, however, has the drawback that the model loses its tractability. For instance, for the case of CES preferences, it is not possible to obtain analytical characterizations for portfolios and asset prices. There are three ways in which one can tackle such a model. First, one can attempt to solve the model numerically. To our knowledge, this has been done only for the complete-markets case (Gourinchas and Rey (2006))—an extension to the incomplete-markets case is a daunting task. Second, one can follow, for example, Devereux and Sutherland (2010) and Tille and van Wincoop (2010) and approximate around a deterministic steady state to derive equilibrium characterizations for small volatilities. Finally, one can recognize that log-linear preferences are a special case of CES preferences and build on our model to find an approximate solution for the CES case. To do so, one can perturb the equilibrium in our economy by expanding around the unitary elasticity of substitution, for which the solutions are analytical.¹⁴ The advantage of this approach is that the approximation is done around a stochastic equilibrium as opposed to a deterministic steady state.

To be able to fit the model to the data and to generate a realistic equity premium, we need to extend our model to include investment. In our framework, there is no difference between savings and the external accounts, and the introduction of investment breaks this tight link. In this paper we have decided to concentrate on the valuation effect while holding capital fixed. In the data, of course, movements in investment play an important role. Since our model does not produce realistic equity premia and some other asset-pricing moments, expected capital gains (anticipated valuation changes) are smaller than those documented by Gourinchas and Rey (2007b). To improve the performance of our model along this dimension, one could integrate an external habit into the preferences' specification. Single-good models with external habits are known to match key asset pricing moments very well (Campbell and Cochrane (1999)). A recent work by Stathopoulos (2008) shows that external habits are also successful in matching standard asset pricing moments in a twocountry two-good model with logarithmic preferences (and external habits), with the tractability offered by a logarithmic utility still being preserved.

Some theoretical properties linking financial assets and external accounts derived in our model are easily testable using new datasets that account for valuation effects (Gourinchas and Rey (2007b), Lane and Milesi-Ferretti (2007)). Tests of these relationships, as well as a calibration of our model are beyond the scope of this paper. A calibration exercise is undertaken in Pavlova and Rigobon (2010), who find that our model is consistent with the data along several dimensions. For example, capital gains on countries' NFA positions are large, volatile, serially uncorrelated, and effectively dominate the movements in NFA (Kollmann (2006) and Lane and Shambaugh (2010)). Furthermore, these capital gains are negatively correlated with the trade balance, and so capital gains stabilize countries' NFA positions by offsetting trade balance movements (Gourinchas and Rey (2007b), Devereux and Sutherland (2010)).

 $^{^{14}}$ This idea is closely related to the works of Judd (1998) and Kogan and Uppal (2003) who develop applications of perturbation methods to solving problems in economics and finance.

7. Conclusion

In his Harms Lecture at the Kiel Institute, Obstfeld (2004) stresses that "recent changes in the functioning of international capital markets require a new view of external adjustment" and moreover, that any notion of "external balance adjustment cannot be defined without reference to the structure of national portfolios." In this paper, we take a step in that direction. We develop an open economy model with endogenous portfolio decisions, in which we investigate the interaction between capital markets and the external adjustment process.

From the methodological point of view, our contribution is to construct a framework that is rich enough to include multiple risky assets, incomplete markets, and supply- and demand-side uncertainty, while at the same time simple enough to allow for closed-form characterizations of asset prices, net foreign asset positions, and equity portfolios. It is within this framework that we are able to establish the interconnections between the real side of the economy represented by the trade balance, current account, and consumption allocations and the financial side such as portfolio holdings, stock prices, and valuation changes.

Appendix

Proof of Proposition 1. In this proof, we closely follow He and Pearson (1991). Their analysis is presented in the context of a single-good economy, but this does not present a difficulty for us because (in the main text) we have reduced our problem to a representation that is equivalent to a familiar single-good one. In particular, the first-order conditions for the consumer problem (11)-(12) have the familiar form

$$e^{-\rho t} \frac{\gamma_i(t)}{\overline{\mathbb{C}}_i(t)} = y_i \xi_{\nu_i}(t), \qquad i \in \{H, F\},$$
(A.1)

where the Lagrange multiplier y_i is such that the budget constraint evaluated at the optimal consumption expenditure, $\overline{\mathbb{C}_i}$, is satisfied with equality:

$$E\left[\int_0^T \xi_{\nu_i}(t)\overline{\mathbb{C}}_i(t)dt\right] = W_i(0), \qquad i \in \{H, F\}$$

It follows that, by no-arbitrage, the time-t wealth of a consumer is given by

$$W_i(t) = E_t \left[\int_t^T \frac{1}{\xi_{\nu_i}(t)} \xi_{\nu_i}(s) \overline{\mathbb{C}}_i(s) ds \right], \qquad i \in \{H, F\},$$

and hence, making use of (A.1) and the assumption that γ_i are martingales, we have

$$W_i(t) = \frac{\gamma_i(t)}{y_i \xi_{\nu_i}(t)} \frac{e^{-\rho t} - e^{-\rho T}}{\rho}, \qquad i \in \{H, F\}.$$
 (A.2)

Of course, for the case of the Foreign country, the argument γ_F is constant over time.

To find optimal portfolios, we apply Itô's lemma to (A.2) and match the diffusion term with that in the dynamic budget constraint (10). This operation yields

$$x_i^{\top}(t)\sigma(t) = \frac{\sigma_{\gamma_i}(t)}{\gamma_i(t)} + (m(t) + \nu_i(t))^{\top}, \qquad (A.3)$$

where we have used equation (14). Recall that in incomplete markets the matrix σ is not a square matrix, and hence the above system of equations contains 4 equations (dimensionality of the vector of Brownian motions) in 2 unknowns (the number of stocks). It has a solution if and only if its right-hand side lies in Span(σ). This entails a restriction

$$(I_4 - \sigma(t)^{\top} (\sigma(t)\sigma(t)^{\top})^{-1} \sigma(t)) \frac{\sigma_{\gamma_i}(t)^{\top}}{\gamma_i(t)} + \nu_i(t) = 0,$$
(A.4)

where we have applied the projection operator $I_4 - \sigma(t)^{\top} (\sigma(t)\sigma(t)^{\top})^{-1}\sigma(t)$ to both sides of (A.3). Equation (17) then follows immediately. Note that, for the case of Foreign, equation (A.4) simplifies to yield $\nu_F(t) = \mathbf{0}$ because σ_{γ_F} is equal to zero.

The optimal portfolios are obtained from (A.3) via simple algebraic manipulations that, in particular, make use of the property that $\sigma(t)\nu_i(t) = 0$.

Proof of Lemma 1. We use the construct of the representative agent to value stocks in the economy. The representative agent's utility evaluated at the aggregate output is given by

$$u(Y(t), Y^{*}(t); \lambda(t)) = \max_{\substack{C_{H}(t) + C_{F}(t) = Y(t), \\ C_{H}^{*}(t) + C_{F}^{*}(t) = Y^{*}(t)}} u_{H}(C_{H}(t), C_{H}^{*}(t)) + \lambda(t)u_{F}(C_{F}(t), C_{F}^{*}(t)).$$

It follows from this definition that the marginal utilities of the representative agent and the individual agents, evaluated at the optimum, are related as

$$\nabla u(Y(t), Y^*(t); \lambda(t)) = \nabla u_H(\overline{C}_H(t), \overline{C}^*_H(t)) = \lambda(t) \nabla u_F(\overline{C}_F(t), \overline{C}^*_F(t)),$$

where the symbol ∇ is used to denote the gradient. From the first-order conditions of the Home consumer,

$$\nabla u_H(\overline{C}_H(t), \,\overline{C}_H^*(t)) = \Big(y_H p(t)\xi_{\nu_H}(t), \, y_H p^*(t)\xi_{\nu_H}(t)\Big).$$

To derive this we used the fact that $\nabla u_H(\overline{C}_H(t), \overline{C}_H^*(t)) = (\gamma_H(t)\theta_H(t)/\overline{C}_H(t), \gamma_H(t)(1-\theta_H(t))/\overline{C}_H^*(t))$ combined with (8) and (A.1). Substituting the sharing rules of the representative agent (18), we can then derive the personalized state price density of the Home consumer and hence that of the representative agent:

$$\xi_{\nu_H}(t) = e^{-\rho t} \frac{p(0)}{p(t)} \, \overline{\overline{C}_H(0)} = e^{-\rho t} \, \frac{p(0)Y(0)}{p(t)Y(t)} \, \frac{\gamma_H(t)\theta_H(t) + \lambda(t)\gamma_F(1-\theta_F)}{\gamma_H(0)\theta_H(0) + \lambda(0)\gamma_F(1-\theta_F)}.$$
(A.5)

This state price density can be used to price the risky stocks by no-arbitrage:¹⁵

$$S(t) = \frac{1}{\xi_{\nu_H}(t)} E_t \left[\int_t^T \xi_{\nu_H}(s) \, p(s) \, Y(s) \, ds \right], \quad S^*(t) = \frac{1}{\xi_{\nu_H}(t)} E_t \left[\int_t^T \xi_{\nu_H}(s) \, p^*(s) \, Y^*(s) \, ds \right].$$

Hence, the price of the Home stock is

$$S(t) = \frac{e^{\rho t} p(t) Y(t)}{\gamma_H(t) \theta_H(t) + \lambda(t) \gamma_F(1 - \theta_F)} E_t \left[\int_t^T e^{-\rho s} (\gamma_H(s) \theta_H(s) + \lambda(s) \gamma_F(1 - \theta_F)) ds \right]$$

$$= \frac{1 - e^{-\rho(T-t)}}{\rho} p(t) Y(t) + \frac{e^{\rho t} \gamma_F(1 - \theta_F) p(t) Y(t)}{\gamma_H(t) \theta_H(t) + \lambda(t) \gamma_F(1 - \theta_F)} E_t \left[\int_t^T e^{-\rho s} (\lambda(s) - \lambda(t)) ds \right], (A.6)$$

where we used the fact that $\gamma_H \theta_H$ is a martingale (i.e., $E_t[\gamma_H(s)\theta_H(s)] = \gamma_H(t)\theta_H(t)$). Analogously, using the fact that $\gamma_H(1 - \theta_H)$ is a martingale, we find the price of the Foreign stock to be

$$S^{*}(t) = \frac{e^{\rho t} p^{*}(t) Y^{*}(t)}{\gamma_{H}(t)(1 - \theta_{H}(t)) + \lambda(t)\gamma_{F}\theta_{F}} E_{t} \left[\int_{t}^{T} e^{-\rho s} (\gamma_{H}(s)(1 - \theta_{H}(s)) + \lambda(s)\gamma_{F}\theta_{F}) ds \right]$$

$$= \frac{1 - e^{-\rho(T-t)}}{\rho} p^{*}(t) Y^{*}(t) + \frac{e^{\rho t} \gamma_{F}\theta_{F} p^{*}(t) Y^{*}(t)}{\gamma_{H}(t)(1 - \theta_{H}(t)) + \lambda(t)\gamma_{F}\theta_{F}} E_{t} \left[\int_{t}^{T} e^{-\rho s} (\lambda(s) - \lambda(t)) ds \right] . (A.7)$$

There are two ways to proceed in evaluating the above conditional expectations. The first is to assume that λ is a martingale (and hence $E_t[\lambda(s) - \lambda(t)] = 0$) and then verify that it is indeed the case in equilibrium. From Proposition 2, however, we can only conclude that λ is a local martingale.

 $^{^{15}}$ All stocks are in positive net supply. The arguments of Loewenstein and Willard (2007) rule out bubbles in the stocks' valuation.

In all special cases that we consider in Section 4, it is easy to verify that λ is also a true martingale under some additional mild regularity conditions imposed on the preference shifts.¹⁶ However, for the general case it is not immediate how to show it.

An alternative approach is to use the following, less direct, argument based on market clearing. In particular, from (A.1)-(A.2), we have

$$W_H(t) + W_F(t) = (\overline{\mathbb{C}}_H(t) + \overline{\mathbb{C}}_F(t)) \frac{1 - e^{-\rho(T-t)}}{\rho} = p(t) Y(t) \frac{1 - e^{-\rho(T-t)}}{\rho} + p^*(t) Y^*(t) \frac{1 - e^{-\rho(T-t)}}{\rho},$$

where in the last equality we used the fact that the total consumption expenditure at time t equals $p(t)Y(t) + p^*(t)Y^*(t)$. On the other hand, from stock market clearing, we have

$$W_H(t) + W_F(t) = S(t) + S^*(t).$$

Combining the resulting restriction that

$$S(t) + S^{*}(t) = p(t) Y(t) \frac{1 - e^{-\rho(T-t)}}{\rho} + p^{*}(t) Y^{*}(t) \frac{1 - e^{-\rho(T-t)}}{\rho}$$

with (A.6)-(A.7), we conclude that

$$S(t) = p(t) Y(t) \frac{1 - e^{-\rho(T-t)}}{\rho}$$
 and $S^*(t) = p^*(t) Y^*(t) \frac{1 - e^{-\rho(T-t)}}{\rho}$.

This establishes (21)-(22).

To derive (23), we combine (A.1)–(A.2) with (18)–(19) and use the representation of the stock prices S and S^* derived in this lemma.

Proof of Lemma 4. Equation (31) follows from $NFA_H(t) = W_H(t) - S(t)$ and Lemma 1. To derive (30), we use the definition of the trade balance, $TB_H(t) = (p(t)(Y(t) - C_H(t)) - p^*(t)Y^*(t))dt$, substitute the equilibrium expressions for consumption and the terms of trade, (18) and (20), and simplify.

Before we proceed to the rest of the proofs, we need to define several auxiliary vectors to be used throughout the remainder of this appendix. Let

$$i_{1} \equiv (1, 0, 0, 0), \quad i_{2} \equiv (0, 1, 0, 0), \text{ and}$$

$$A(t) \equiv \frac{\gamma_{H}(t)\sigma_{\gamma_{H}}(t) + \theta_{H}(t)\sigma_{\theta_{H}}(t) - \gamma_{F}(1 - \theta_{F})\lambda(t)\nu_{H}(t)^{\top}}{\gamma_{H}(t)\theta_{H}(t) + \lambda(t)\gamma_{F}(1 - \theta_{F})} - \frac{(1 - \theta_{H}(t))\sigma_{\gamma_{H}}(t) - \gamma_{H}\sigma_{\theta_{H}}(t) - \gamma_{F}\theta_{F}\lambda(t)\nu_{H}(t)^{\top}}{\gamma_{H}(t)(1 - \theta_{H}(t)) + \lambda(t)\gamma_{F}\theta_{F}} - \sigma_{Y}(t)i_{1} + \sigma_{Y^{*}}(t)i_{2}.$$
(A.9)

¹⁶The only special case that requires these additional assumptions is that presented in Section 4.3. In particular, one needs to bound the preference shifts in such a way that the expression in (37) satisfies $E\left[e^{\frac{1}{2}\int_0^T \nu_H(t)^\top \nu_H(t)dt}\right] < \infty$. This condition is known as the Novikov condition.

Proof of Proposition 2. By substituting (A.2) into (24) we derive

$$\lambda(t) = \frac{y_H \,\xi_{\nu_H}(t)}{y_F \,\xi_{\nu_F}(t)}.$$

Applying Itô's lemma and using the representation of the countries' state price densities from (14), we have

$$d\lambda(t) = -\lambda(t)m(t)^{\top}\nu_{H}(t)dt - \lambda(t)\nu_{H}(t)d\vec{w}(t), \qquad (A.10)$$

where we have substituted the finding that $\nu_F(t) = \mathbf{0}$ established in Proposition 1. To show that the drift term in (A.10) is equal to zero, we use the definition of m from (13) and the restriction that $\sigma(t)\nu_H(t) = 0$.

To determine $\lambda(0)$, note from Lemma 1 that the initial financial wealth of, say, the Home country is given by (23) evaluated at t = 0. On the other hand, $W_H(0) = S(0)$ because the initial portfolio of Home consists of one share of the Home stock. This allows us to pin down $\lambda(0)$. It is easy to show that $\lambda(0) = \gamma_H(0)(1 - \theta_H(0))/\gamma_F(1 - \theta_F)$.

We now report the volatility matrix of stock returns.

$$\sigma(t) = \begin{bmatrix} \frac{1-a}{a q(t)+1-a} A(t) + \sigma_Y(t) i_1 \\ -\frac{a q(t)}{a q(t)+1-a} A(t) + \sigma_{Y^*}(t) i_2 \end{bmatrix},$$

where A(t), i_1 , and i_2 are defined in (A.8)–(A.9). This volatility matrix is obtained by applying Itô's lemma to the closed-form expressions for the stock prices (21)–(22).

The market price of risk process m can be derived from the dynamics of ξ_H in (14). Using the identity $\xi_H(t) = ap(t)\xi_H(t) + (1-a)p^*(t)\xi_H(t)$ and equations (20) and (A.5), we derive

$$\xi_{H}(t) = ae^{-\rho t}p(0)\frac{\gamma_{H}(t)\theta_{H}(t) + \lambda(t)\gamma_{F}(1-\theta_{F})}{\gamma_{H}(0)\theta_{H}(0) + \lambda(0)\gamma_{F}(1-\theta_{F})}\left(\frac{Y(0)}{Y(t)}\right) + (1-a)e^{-\rho t}p(0)\frac{\gamma_{H}(t)(1-\theta_{H}(t)) + \lambda(t)\gamma_{F}\theta_{F}}{\gamma_{H}(0)\theta_{H}(0) + \lambda(0)\gamma_{F}(1-\theta_{F})}\left(\frac{Y(0)}{Y^{*}(t)}\right).$$
(A.11)

Applying Itô's lemma and identifying the diffusion term with that in the representation of ξ_H in (14), we obtain

$$\begin{split} m(t) &= -\frac{p(0)e^{-\rho t}}{\xi_{H}(t)} \frac{Y(0)}{\gamma_{H}(0)\theta_{H}(0) + \lambda(0)\gamma_{F}(1-\theta_{F})} \left[a \, \frac{\gamma_{H}(t)\sigma_{\gamma_{H}}(t) + \theta_{H}(t)\sigma_{\theta_{H}}(t) - \gamma_{F}(1-\theta_{F})\lambda(t)\nu_{H}(t)}{Y(t)} \right. \\ &+ (1-a) \, \frac{(1-\theta_{H}(t))\sigma_{\gamma_{H}}(t) - \gamma_{H}\sigma_{\theta_{H}}(t) - \gamma_{F}\theta_{F}\lambda(t)\nu_{H}(t)}{Y^{*}(t)} \\ &- a \, \frac{\gamma_{H}(t)\theta_{H}(t) + \gamma_{F}(1-\theta_{F})\lambda(t)}{Y(t)}\sigma_{Y}(t)i_{1} - (1-a) \, \frac{\gamma_{H}(t)(1-\theta_{H}(t)) + \gamma_{F}\theta_{F}\lambda(t)}{Y^{*}(t)}\sigma_{Y^{*}}(t)i_{2} \right] - \nu_{H}(t). \end{split}$$

This completes the proof of the proposition. \blacksquare

Derivation of the remaining equilibrium quantities. We first report the interest rate r and the stocks' expected returns μ_s and μ_{s^*} and then explain how we derived these expressions.

$$r(t) = \rho + \frac{aq(t)}{aq(t) + 1 - a} \left(\mu_{Y}(t) - \sigma_{Y}(t)^{2} \right) + \frac{1 - a}{aq(t) + 1 - a} \left(\mu_{Y^{*}}(t) - \sigma_{Y^{*}}(t)^{2} \right) + \frac{aq(t)}{aq(t) + 1 - a} \frac{\sigma_{Y}(t)i_{1}(\gamma_{H}(t)\sigma_{\theta_{H}}(t)^{\top} + \theta_{H}(t)\sigma_{\gamma_{H}}(t)^{\top} - \lambda(t)\nu_{H}(t)\gamma_{F}(1 - \theta_{F})))}{\gamma_{H}(t)\theta_{H}(t) + \lambda\gamma_{F}(1 - \theta_{F})(t)} + \frac{1 - a}{aq(t) + 1 - a} \frac{\sigma_{Y^{*}}(t)i_{2}(-\gamma_{H}(t)\sigma_{\theta_{H}}(t)^{\top} + (1 - \theta_{H}(t))\sigma_{\gamma_{H}}(t)^{\top} - \lambda(t)\nu_{H}(t)\gamma_{F}\theta_{F})}{\gamma_{H}(t)(1 - \theta_{H}(t)) + \lambda\gamma_{F}\theta_{F}},$$
(A.12)

$$\mu_{S}(t) = \rho + \mu_{Y}(t) + \frac{1-a}{aq(t)+1-a} \left(\mu_{q}(t) - \frac{aq(t)}{aq(t)+1-a} ||A(t)||^{2} + \sigma_{Y}(t)A(t)i_{1}^{\top} \right),$$
(A.13)

$$\mu_{S^*}(t) = \rho + \mu_{Y^*}(t) + \frac{aq(t)}{aq(t) + 1 - a} \left(-\mu_q(t) + \frac{aq(t)}{aq(t) + 1 - a} ||A(t)||^2 - \sigma_{Y^*}(t)A(t)i_2^\top \right), \quad (A.14)$$

where μ_q is the expected improvement in the terms of trade, given by

$$\mu_{q}(t) = \mu_{Y^{*}}(t) - \mu_{Y}(t) + \frac{1}{2} ||A(t)||^{2} - \frac{1}{2} \frac{||\gamma_{H}(t)\sigma_{\theta_{H}}(t)^{\top} + \theta_{H}(t)\sigma_{\gamma_{H}}(t)^{\top} - \lambda(t)\nu_{H}(t)\gamma_{F}(1-\theta_{F})||^{2}}{(\gamma_{H}(t)\theta_{H}(t) + \lambda\gamma_{F}(1-\theta_{F})(t))^{2}} + \frac{1}{2} \frac{||-\gamma_{H}(t)\sigma_{\theta_{H}}(t)^{\top} + (1-\theta_{H}(t))\sigma_{\gamma_{H}}(t)^{\top} - \lambda(t)\nu_{H}(t)\gamma_{F}\theta_{F}||^{2}}{(\gamma_{H}(t)(1-\theta_{H}(t)) + \lambda\gamma_{F}\theta_{F}(t))^{2}} + \frac{1}{2}\sigma_{Y}(t)^{2} - \frac{1}{2}\sigma_{Y^{*}}(t)^{2},$$

and where A(t), i_1 , and i_2 are defined in (A.8)–(A.9).

The interest rate r in (A.12) is equal to the drift term from the Itô expansion of the equilibrium state price density reported in (A.11). The formulas in (A.13)–(A.14) are obtained by applying Itô's lemma to the closed-form expressions for the stock prices (21)–(22) and the terms of trade (20), and then using the definitions of μ_s and μ_{s^*} from (4)–(5). Throughout these derivations, we have used our assumption that γ_H and $\theta_H \gamma_H$ are martingales.

Derivations for Section 4. All derivations for the special cases examined in Section 4 are tedious but straightforward. Perhaps the easiest way to obtain the formulas and signs reported in that section is to use *Mathematica* to simplify the expressions derived above and manipulate them in *Mathematica* to verify the desired properties. Our programs are available upon request.

References

- ADLER, M., AND B. DUMAS (1983): "International Portfolio Choice and Corporation Finance: a Synthesis," *Journal of Finance*, 38, 925–984.
- BASAK, S., AND B. CROITORU (2000): "Equilibrium Mispricing in a Capital Market with Portfolio Constraints," *Review of Financial Studies*, 13, 715–748.
- BASAK, S., AND D. CUOCO (1998): "An Equilibrium Model with Restricted Stock Market Participation," *Review of Financial Studies*, 11(2), 309–341.
- BLANCHARD, O., F. GIAVAZZI, AND F. SA (2005): "International Investors, the U.S. Current Account, and the Dollar," *Brookings Papers on Economic Activity*, 1, 1–65.
- BREEDEN, D. T. (1979): "An Intertemporal Asset Pricing Model with Stochastic Consumption and Investment Opportunities," *Journal of Financial Economics*, 7(1), 265–296.
- CABALLERO, R. J., E. FARHI, AND P.-O. GOURINCHAS (2008): "An Equilibrium Model of "Global Imbalances" and Low Interest Rates," *American Economic Review*, 98, 358–393.
- CAMPBELL, J. Y., AND J. H. COCHRANE (1999): "By Force of Habit: A Consumption-Based Explanation of Aggregate Stock Market Behavior," *Journal of Political Economy*, 107, 205–251.
- CASS, D., AND A. PAVLOVA (2004): "On Trees and Logs," Journal of Economic Theory, 116, 41–83.
- COLE, H. L., AND M. OBSTFELD (1991): "Commodity Trade and International Risk Sharing," Journal of Monetary Economics, 28, 3–24.
- CUOCO, D., AND H. HE (1994): "Dynamic Equilibrium in Infinite-Dimensional Economies with Incomplete Financial Markets," working paper, University of Pennsylvania.
- DEN HAAN, W., K. JUDD, AND M. JUILLARD (2010): "Computational Suite of Models with Heterogenous Agents: Multi-Country Real Business Cycle Models," *Journal of Economic Dynamics and Control*, 34, 1–3.
- DETEMPLE, J. B., AND A. SERRAT (2003): "Dynamic Equilibrium with Liquidity Constraints," *Review of Financial Studies*, 16, 597–629.
- DEVEREUX, M. B., AND M. SAITO (2006): "A Portfolio Theory of International Capital Flows," working paper, University of British Columbia.
- DEVEREUX, M. B., AND A. SUTHERLAND (2010): "Valuation Effects and the Dynamics of Net External Assets," *Journal of International Economics*, 80, 129–143.
- (2011): "Country Portfolios in Open Economy Macro-Models," Journal of the European Economic Association, 9, 337–369.
- DOOLEY, M. P., D. FOLKERTS-LANDAU, AND P. GARBER (2004): "The Revised Bretton Woods System," International Journal of Finance and Economics, 9(4), 307–313.
- DORNBUSCH, R., S. FISCHER, AND P. A. SAMUELSON (1977): "Comparative Advantage, Trade, and Payments in a Ricardian Model with a Continuum of Goods," *American Economic Review*, 67, 823–839.
- ENGEL, C., AND A. MATSUMOTO (2009): "The International Diversification Puzzle when Goods Prices are Sticky: Its Really about Exchange-Rate Hedging, not Equity Portfolios," *American Economic Journal: Macroeconomics*, 1(2), 155–88.
- EVANS, M., AND V. HNATKOVSKA (2007): "Solving General Equilibrium Models with Incomplete Markets and Many Financial Assets," working paper, Georgetown University.

- GHIRONI, F., J. LEE, AND A. REBUCCI (2006): "The Valuation Channel of External Adjustment," working paper, Boston University.
- GOURINCHAS, P.-O., N. GOVILLOT, AND H. REY (2010): "Exorbitant Privilege and Exorbitant Duty," working paper, London Business School.
- GOURINCHAS, P.-O., AND H. REY (2006): "Intertemporal Approach to the Financial Account," work in progress, University of California, Berkeley.

(2007a): "From World Banker to World Venture Capitalist: The US External Adjustment and The Exorbitant Privilege," in "G7 Current Account Imbalances: Sustainability and Adjustment," R. Clarida, ed., The University of Chicago Press, 11–55.

(2007b): "International Financial Adjustment," *Journal of Political Economy*, 115, 665–703.

- HE, H., AND N. D. PEARSON (1991): "Consumption and Portfolio Policies with Incomplete Markets and Short-Sale Constraints: The Infinite Dimensional Case," *Journal of Economic Theory*, 54, 259–304.
- HELPMAN, E., AND A. RAZIN (1978): A Theory of International Trade under Uncertainty. Academic Press, San Diego.
- JUDD, K. L. (1998): Numerical Methods in Economics. MIT Press, Cambridge, MA.
- JUDD, K. L., AND S.-M. GUU (2001): "Asymptotic Methods for Asset Market Equilibrium Analysis," *Economic Theory*, 18, 127–157.
- KEHOE, P. J., AND F. PERRI (2002): "International Business Cycles with Endogenous Incomplete Markets," *Econometrica*, 70, 907–928.
- KOGAN, L., AND R. UPPAL (2003): "Risk Aversion and Optimal Portfolio Policies in Partial and General Equilibrium Economies," CEPR Discussion Paper 3306.
- KOLLMANN, R. (2006): "International Portfolio Equilibrium and the Current Account," working paper, ECARES, Free University of Brussels.
- KOURI, P. (1982): "Balance of Payment and the Foreign Exchange Market: A Dynamic Partial Equilibrium Model," in Jagdeep Bhandari and Bluford Putnam, eds., *Economic Interdependance* and Flexible Exchange Rates, (MIT Press), 116-156.
- KRAAY, A., AND J. VENTURA (2000): "Current Accounts in Debtor and Creditor Countries," *Quarterly Journal of Economics*, 115, 1137–1166.
 - (2003): "Current Accounts in the Long and Short Run," *NBER Macroeconomics Annual* 2002, 17, 65–94.
- LANE, P. R., AND G. M. MILESI-FERRETTI (2001): "The External Wealth of Nations: Measures of Foreign Assets and Liabilities for Industrial and Developing Countries," *Journal of International Economics*, 55, 263–294.

(2007): "The External Wealth of Nations Mark II: Revised and Extended Estimates of Foreign Assets and Liabilities, 1970-2004," *Journal of International Economics*, 73, 223–250.

- LANE, P. R., AND J. C. SHAMBAUGH (2010): "Financial Exchange Rates and International Currency Exposures," American Economic Review, 100, 518–540.
- LOEWENSTEIN, M., AND G. WILLARD (2007): "Consumption and Bubbles," working paper, University of Maryland.
- MAGGIORI, M. (2012): "Financial Intermediation, International Risk Sharing, and Reserve Currencies," working paper, Harvard University.

- MARCET, A., AND R. MARIMON (1999): "Recursive Contracts," working paper, Universitat Pompeu Fabra.
- MARKOWITZ, H. (1952): "Portfolio Selection," Journal of Finance, 7, 77–91.
- MERTON, R. C. (1971): "Optimum Consumption and Portfolio Rules in a Continuous-Time Model," *Journal of Economic Theory*, 3, 373–413.
- NEGISHI, T. (1960): "Welfare Economics and Existence of an Equilibrium for Competitive Economy," *Metronomica*, 12, 92–97.
- OBSTFELD, M. (2004): "External Adjustment," Review of World Economics, 140, 541–568.
- OBSTFELD, M., AND K. S. ROGOFF (1996): Foundations of International Macroeconomics. MIT Press, Massachusetts.
- PAVLOVA, A., AND R. RIGOBON (2007): "Asset Prices and Exchange Rates," *Review of Financial Studies*, 20, 1139–1181.
- (2008): "The Role of Portfolio Constraints in the International Propagation of Shocks," *Review of Economic Studies*, 75, 1215–1256.
- (2010): "An Asset-Pricing View of External Adjustment," Journal of International Economics, 80, 144–156.
- SAMUELSON, P. A. (1970): "The Fundamental Approximation Theorem of Portfolio Analysis in terms of Means, Variances and Higher Moments," *Review of Economic Studies*, 37, 537–542.
- SCHRODER, M., AND C. SKIADAS (2003): "Optimal Lifetime Consumption-Portfolio Strategies under Trading Constraints and Generalized Recursive Preferences," *Stochastic Processes and their Applications*, 108, 155–202.
- STATHOPOULOS, A. (2008): "Asset Prices and Risk Sharing in Open Economies," working paper, Columbia University.
- STOCKMAN, A. C., AND L. L. TESAR (1995): "Tastes and Technology in a Two-Country Model of the Business Cycle: Explaining International Comovements," *American Economic Review*, 85, 168–185.
- TILLE, C. (2003): "The Impact of Exchange Rate Movements on U.S. Foreign Debt," Federal Reserve Bank of New York Current Issues in Economics and Finance 9.
- ——— (2008): "Financial Integration and the Wealth Effect of Exchange Rate Fluctuations," Journal of International Economics, 75, 283–294.
- TILLE, C., AND E. VAN WINCOOP (2010): "International Capital Flows," Journal of International Economics, 80, 157–175.
- ZAPATERO, F. (1995): "Equilibrium Asset Prices and Exchange Rates," Journal of Economic Dynamics and Control, 19, 787–811.